



No.1854 June 2022

Foreign ownership and robot adoption

Fabrizio Leone



THE LONDON SCHOOL OF ECONOMICS AND POLITICAL SCIENCE



Economic and Social Research Council

Abstract

This paper shows that multinational enterprises (MNEs) spur the adoption of industrial robots. First, I document a positive and robust correlation between multinational production and robot adoption using a new cross-country industry-level panel. Second, using detailed data about Spanish manufacturing, I combine a difference-in-differences approach with a propensity score reweighing estimator and provide evidence that firms switching from domestic to foreign ownership become about 10% more likely to employ robots. The ability of expanding into foreign markets via the parental network is the key driver of the adoption choice. An empirical model of firm investment reveals that MNEs generate significant industry-level productivity gains but decreases the labor share by boosting robot adoption. However, the first effect is one order of magnitude larger than the second. These results provide new evidence about the efficiency versus equity trade-off that policymakers face when attracting MNEs.

Keywords: foreign ownership, industrial robots, total factor productivity, factor-biased productivity, labor share JEL: F23; O33

This paper was produced as part of the Centre's Trade Programme. The Centre for Economic Performance is financed by the Economic and Social Research Council.

An earlier version of this paper was titled "Multinational Enterprises, Technology Transfers, and Robot Adoption". I thank Paola Conconi and Mathieu Parenti for their guidance and support. I thank Ludovica Gazze, Rocco Macchiavello, Glenn Magerman, Guy Michaels, Tristan Reed, Catherine Thomas, John Van Reenen, and Jose Vasquez for useful comments and seminar participants at the ECARES lunch seminar, CIE Conference, XXXV Jornadas de Economía Industrial, Young Economist Symposium, ENTER Jamboree, European Trade Study Group, RIEF Doctoral Meeting, University College of London ENTER Seminar, KIIS Workshop, and XIV FIW Conference in International Economics. All errors are mine.

Fabrizio Leone, ECARES, Université Libre de Bruxelles and visiting PhD student at the Centre for Economic Performance, London School of Economics, 2022.

Published by Centre for Economic Performance London School of Economics and Political Science Houghton Street London WC2A 2AE

All rights reserved. No part of this publication may be reproduced, stored in a retrieval system or transmitted in any form or by any means without the prior permission in writing of the publisher nor be issued to the public or circulated in any form other than that in which it is published.

Requests for permission to reproduce any article or part of the Working Paper should be sent to the editor at the above address.

© F. Leone, submitted 2022.

1 Introduction

Multinational enterprises (MNEs) are crucial players in the world economy. They shape global production through Foreign Direct Investment (FDI), are responsible for about two-thirds of international trade flows (UNCTAD, 2013), and are important transnational employers. From the host countries' standpoint, MNEs are frequently praised for bringing superior technology (Harrison and Rodríguez-Clare, 2010). Indeed, they often employ more innovative production methods and more efficient management procedures than domestic companies (Bloom, Genakos, Sadun and Van Reenen, 2012). For this reason, foreign-owned firms tend to be bigger and more productive. Governments hope that this knowledge may also flow to domestic-owned firms directly or indirectly linked to a multinational. If this happens, the potential benefits of MNEs' local presence amplify in a substantial way (Javorcik, 2004; Keller, 2021).¹

However, since technological change is typically factor-biased (Doraszelski and Jaumandreu, 2018), attracting foreign investment may deliver not only aggregate gains but also induce distributional outcomes. In this paper, I show that MNEs spur the adoption of industrial robots.² This finding is new in the literature, and it raises potential distributional concerns: unlike standard capital-intensive technology, robots may increase productivity but replace workers in routine production tasks, and therefore pose an efficiency versus equity trade-off for policymakers.³

I exploit two data sources. The first is a new cross-country industry-level dataset containing the number of deployed robots and gross output produced by foreign-owned firms (which I call "multinational production") for 37 countries and 20 industries from 2005 to 2014. Using these data, I document that a one per cent multinational production increase correlates with 0.5% more robots per thousand employees, which is a standard measure of robots' diffusion (Graetz and Michaels, 2018; Acemoglu and Restrepo, 2020). The correlation is significant and robust to the inclusion of observed time-varying country-industry characteristics (e.g., capital-to-labor employment and wage ratio) and any time-varying country and industry-level shocks. I also show that this result is not driven by the top five robot markets (i.e., China, Germany, Japan, South Korea, and the US) or the automotive industry, which accounts alone for over

¹These spillovers provide a potential rational for policy intervention. Typical policies to attract foreign investors include the creation of Investment Promotion Agencies (IPAs), Special Economic Zones, and tax breaks or subsidies (UNCTAD, 2018).

²Industrial robots are defined as "automatically controlled, reprogrammable, multipurpose manipulator programmable in three or more axes, which can be either fixed in place or mobile for use in industrial automation applications" (ISO 8372:2012). See International Federation of Robotics (2019).

³The difference between robots and standard capital inputs has been theoretically and empirically emphasized in a number of papers (Acemoglu and Restrepo, 2018b; Graetz and Michaels, 2018; Acemoglu and Restrepo, 2020; Koch, Manuylov and Smolka, 2021).

80% of the worldwide robot stock in any sample year.

Multiple explanations may underlie the cross-country evidence. On the one hand, the results are consistent with multinational production increasing robot adoption. On the other, reverse causality and omitted factors preclude a causal interpretation. To make further progress in understanding whether foreign ownership boosts robot adoption, I use the firm-level ESEE survey data provided by the SEPI Foundation in Madrid. The data, which cover from 1990 to 2017, are a detailed and representative panel of the population of Spanish manufacturing firms and, crucially for this paper, report the share of firms' equity owned by foreign companies (i.e., headquartered outside of Spain) and a binary variable for whether or not a firm employs robots in a given year. I restrict the sample to two groups of firms. The first group includes firms that stay under domestic ownership throughout their life span. The second group contains firms that switch at most once from domestic to foreign ownership. Although firms belonging to the second group are only about 3% of the total, they account for a disproportionately large share of production and employment, are more innovative, and engage more in international trade than domestic-owned ones.⁴

The data reveal that even if robot adoption rates have steadily grown over time, the (unconditional) share of robot adopters among foreign-owned firms is twice as large than among domestic-owned ones in any year. This pattern holds across different manufacturing sectors and, although lower in magnitude, also after controlling for firm size (e.g., total sales). Next, I introduce an event-study specification to study how the probability of employing robots changes after switching from domestic to foreign ownership. When performing this exercise, I restrict my attention to ever acquired firms.⁵ The identifying assumption is that not-yet-acquired firms are a credible counterfactual for acquired ones after controlling for firm-level time invariant heterogeneity, industry-level time-varying common shocks to all firms, and firm-level time-varying observed differences in levels (e.g., sales) and performance (e.g., sales growth rate). This specification reveals that, as of the first year after the acquisition, new affiliates are about 35% more likely to employ robots than in the year before the event. I find no evidence of anticipation effects.⁶

⁴Guadalupe, Kuzmina and Thomas (2012) and Koch and Smolka (2019) also document this pattern using the ESEE data. The fact that (future) foreign-owned firms are a positively selected sub-sample is also documented beyond the Spanish context (Setzler and Tintelnot, 2021; Alfaro-Ureña, Manelici and Vasquez, forthcoming).

⁵In a robustness, I obtain a similar pattern when including never acquired firms in the control group.

⁶Recent literature has shown that estimating event studies with a two-way fixed effects estimator fail to recover the treatment effect when treatment roll-out is staggered and the treatment effects change over time (Sun and Abraham, 2021; Borusyak, Jaravel and Spiess, 2021; Callaway and Sant'Anna, 2021; Goodman-Bacon, 2021). To deal with this concern, I also estimate the event study using the method proposed by Sun and Abraham (2021). The results are qualitatively similar to the baseline

Although suggestive of a causal relationship, one may worry that the estimates are confounded by unobserved time-varying firm-level shocks that simultaneously influence the foreign acquisition decision and robot adoption (Blundell and Costa Dias, 2009). Absent firm-level experimental variation in equity share composition, I follow the approach of Guadalupe et al. (2012) and combine a difference-in-differences approach with an inverse probability reweighing (IPRW) estimator to address this concern.

The estimator works in two steps. First, for each firm in the sample, I estimate the probability of becoming foreign-owned as a function of an extensive set of controls in levels and growth rates. Second, I assign a weight of $1/\hat{p}$ to acquired firms and $1/(1-\hat{p})$ to always domestic ones, being \hat{p} the estimated first-stage probability. This reweighing scheme aims at making the two groups of firms comparable on observables before the acquisition. Then, I project a binary variable for whether a firm employs a robot in a given year on a binary variable for whether it is foreign-owned in that year and firm and industry-year fixed effects. The identifying assumption is that, post-reweighing and conditional on the included fixed effects, treatment is random. Using this procedure, I find that, after the acquisition, firms are on average 10% more likely to employ robots than comparable domestic producers, on average.⁷ I use this estimate as my baseline and call it the "foreign ownership robot adoption premium". I also provide evidence that the premium is larger in industries characterized by lower adoption rates at the beginning of the sample period.

Various mechanisms may explain why being acquired by a foreign multinational spurs robot adoption. For instance, using the same data, previous literature has shown that the possibility of expanding into foreign markets via the parental network is a central driver of process and product innovation (Guadalupe et al., 2012) and skill-upgrading (Koch and Smolka, 2019). To the extent that robots are complementary to these activities, the increased foreign market access channel is also relevant in my context. An alternative hypothesis is that foreign acquisitions alleviate credit constraints, thus allowing firms to move closer to their first-best investment level (Harrison and McMillan, 2003; Manova, Wei and Zhang, 2015). Another possible explanation is that foreign parents directly transfer robot adoption knowledge to their affiliates. Indeed, previous literature has found extensive evidence of technology flows within multinational networks (Branstetter, Fisman and Foley, 2006; Keller and Yeaple, 2013; Bilir and Morales, 2020).

The richness of the ESEE survey allows me to test the explanatory power of these alternative mechanisms. To evaluate the foreign market access channel, I use a ques-

specification.

⁷By contrast, I find no evidence that acquired firms become more likely to use other automation-type non-robotic technologies (e.g., numerically controlled machines) that are older and less productivity-enhancing than robots (Acemoglu and Restrepo, 2020; Koch et al., 2021).

tion in the survey that asks firms whether they access export markets (if at all) via their foreign parental network or via other means. To assess the importance of credit constraints, I use information on firm expenditure on external R&D, an activity heavily dependent on firms' borrowing possibilities (Brown, Martinsson and Petersen, 2012). Finally, although the ESEE does not contain information about intra-firm activities, I proxy technology transfers with expenses on licenses and technical aid from abroad, possibly from the parent firm.

Using the same approach as in the baseline, I find that switching to foreign ownership boosts affiliates' foreign market access which, in turn, is a driver of robot adoption. Intuitively, acquired firms need to scale-up production to translate higher potential demand into actual sales, and robot adoption is one way to achieve this result. The estimates also reveal that the effect is stronger for newly acquired firms that did not export before the acquisition year. By contrast, I find no support for the other two hypotheses.

One interesting question is whether different types of acquisitions, e.g., from a domestic firm, produce different robot adoption outcomes. Providing an answer is relevant from a policy perspective because it reveals whether attracting foreign investment may deliver different outcomes than policies promoting, for instance, domestic mergers.⁸ To do so, I exploit a survey question that asks firms if they are acquired by another domestic company. Using a similar strategy as in the baseline analysis, I provide evidence that firms acquired by a Spanish (non-multinational) company do not become more likely to employ robots. In the light of the previous discussion, one potential reason is that, by construction, companies only active in Spain do not offer a global network of contacts that their affiliates can leverage to expand in foreign markets and, therefore, do not provide sufficient incentives to increase production.

Since robot adoption allows to scale-up production and expand in global markets, one expects to find large within-firm gains associated with ownership switching. Indeed, I find that acquired firms increase sales and employment by about 37% and 22% respectively. However, switching to foreign ownership also redistributes income away from labor. Although wages increase by about 9%, the labor share in revenues falls post-acquisition, suggesting that firms tilt production towards capital-intensive activities.⁹

These insights allow me to parametrize a dynamic partial equilibrium firm invest-

⁸Another question is whether firms acquired by a multinational headquartered in Spain are also more likely to employ robots. In principle, given the documented importance of increased market access through the parental network, I expect that being acquired by a Spanish multinational produces similar outcomes as being acquired by a foreign one. Unfortunately, the ESEE survey does not report if a firm is acquired by a Spanish multinational, which makes it unfeasible to evaluate this hypothesis.

⁹This result is consistent with Koch and Smolka (2019) and Sun (2020).

ment model based on Doraszelski and Jaumandreu (2018), which I use to further analyze the gains and distributional implications of foreign ownership and robot adoption. In the model, firms combine physical inputs and labor with constant elasticity of substitution and make robot adoption choices to maximize profits. Firms' production function features labor-augmenting and total factor productivity (TFP), and I let both depend on past ownership structure and robot adoption choices. The main estimands of interest are the impact of foreign ownership and robot adoption on total factor and labor-augmenting productivity, which governs the physical inputs-to-labor ratio at the firm level. As standard in the literature (Doraszelski and Jaumandreu, 2013, 2018), I assume that productivity follows a first-order Markov process and, therefore, past ownership structure and robot adoption choices are mean independent of current idiosyncratic productivity shocks.

The results reveal the ratio of physical inputs to labor is about 40% larger in robot adopters than among non-adopting firms. After controlling for robot adoption, being foreign-owned does not directly alter firms' optimal input mix. Hence, the model estimates suggest that it is robot adoption, rather than foreign ownership per se, to be the main driver of increased within-firm inequality post-acquisition. By contrast, robot adoption and foreign ownership are associated with 29%, respectively 43%, higher TFP. I use these results to simulate two counterfactual scenarios. First, I simulate an economy without multinationals. When doing so, I account for the reduced-form effect that foreign ownership induces on robot adoption. Second, I consider a scenario without multinationals and robots. In each case, I start from the observed productivity distribution and simulate forward firm-specific total factor and labor-augmenting productivity paths using the estimated law of motions and shutting down one channel at a time.¹⁰

As a benchmark, the model-implied TFP and labour share total growth rates over the sample period are 0.60% and -18.07%, respectively. These figures are qualitatively similar to Doraszelski and Jaumandreu (2018). The counterfactual exercises suggest that an economy without multinationals and robots would have experienced negative total TFP growth (-0.50%) but a higher labour share (1.22%). Shutting down only multinational activity explains about 77% of the total TFP drop but only about 7% of the total labour share increase between scenarios. In each case, robots account for the remaining gap. Overall, the counterfactual exercises deliver two main insights. First, multinational activity produces significant TFP gains but puts downward pressure on the labor share by boosting robot adoption. However, the first effect is one order of

¹⁰To avoid dealing with sample selection, I perform the counterfactuals using the sub-sample of firms always active during the sample period. While the share of foreign-owned firms is about the same as in the original sample, the share of robot adopters is about twice as large. Hence, the results should be understood as an upper bound of the effects in the complete sample.

magnitude larger than the second. Although the model is silent about total welfare, this result offers new evidence about the efficiency versus equity trade-off that policymakers face when attracting foreign investment. Second, the model results also provide a new rationale for the observed decline in the manufacturing labor share across many countries. Grossman and Oberfield (forthcoming) include globalization and automation among the leading forces behind this phenomenon. My results reinforce and extend their argument. Rather than being different explanations, MNEs' activities (globalization) and robot adoption (technological change) jointly contribute to the observed decline. However, robots play a predominant role.

I organize the rest of the paper as follows. Section 2 connects this paper with previous literature. Section 3 introduces the data. Section 4 analyzes the correlation between MNEs' activities and robot adoption using cross-country industry-level data. Section 5 presents the baseline estimates of the foreign ownership robot adoption premium. Section 6 contains the model and the counterfactual simulations. Finally, Section 7 concludes.

2 Related Literature

This paper contributes to three strands of the literature. First, it provides new evidence about the technological impact of MNEs on the domestic economy. Second, it participates to the debate about the interplay of globalization and technological change. Finally, it adds to recent research about the effects of automation.

The first strand of the literature shows that multinationals boost host country productivity by promoting investment and technology upgrading (Aitken and Harrison, 1999; Javorcik, 2004; Blalock and Gertler, 2008; Harrison and Rodríguez-Clare, 2010; Guadalupe et al., 2012; Bloom et al., 2012; Bircan, 2019; Méndez-Chacón and Van Patten, 2021). However, the literature also acknowledges that this process is often skillbiased or factor-specific (Feenstra and Hanson, 1997; Koch and Smolka, 2019; Sun, 2020; Setzler and Tintelnot, 2021; Alfaro-Ureña, Manelici and Vasquez, 2021).

My contribution to this literature is twofold. First, I show that MNEs spur the adoption of industrial robots. Far from being standard capital-augmenting technology, they are labor-replacing technology and produce substantial societal impact, spanning labor markets (Graetz and Michaels, 2018; Acemoglu, Lelarge and Restrepo, 2020; Acemoglu and Restrepo, 2020; Humlum, 2021), international trade (Artuc, Paulo and Rijkers, 2018; Stapleton and Webb, 2020), public finance (Freeman, 2015), and electoral outcomes (Anelli, Colantone and Stanig, 2019).

Second, industrial robots are a directly measurable technology. Rather than inferring the technological content of FDI by comparing pre and post-acquisition firms' activities (e.g., changes in sales, R&D expenditures, or workforce composition), the ESEE data offer a direct measure of robotic technology adoption.

Showing that MNEs spur robot adoption is also informative about how globalization and technological change jointly shape the economy. On the one hand, globalization and technological change can lift millions of people out of poverty (Harrison, 2007). On the other, these two phenomena can fundamentally change societies. Moreover, as long as the benefits of globalization and technological change accrue to large, already more productive firms, the economic integration gains are hampered (Brynjolfsson and McAfee, 2014; Baldwin, 2019; Korinek and Stiglitz, 2021). This process can lead indeed to market power concentration, lower innovation rates and higher inequality (Autor, Dorn, Katz, Patterson and Van Reenen, 2020; De Loecker, Eeckhout and Unger, 2020). Quantifying the role of multinational enterprises in this process is key to designing informed policies towards foreign investment.

Finally, this paper contributes to the literature on automation, in particular robot adoption. First, I show how to identify and estimate the impact of robot adoption (and foreign ownership) on factor-biased and Hicks-neutral firm-level productivity. By doing so, I contribute to the literature about factor-bias technological change estimation (Doraszelski and Jaumandreu, 2013, 2018; Raval, 2019; Oberfield and Raval, 2021). Second, by showing that robot adoption boosts productivity but decreases the labor share, I add to the debate about the effects of automation at the firm-level (Acemoglu et al., 2020; Aghion, Antonin, Bunel and Jaravel, 2020; Koch et al., 2021; Humlum, 2021) and industry-level (Graetz and Michaels, 2018; Acemoglu and Restrepo, 2020; Hémous and Olsen, 2022).

3 Data

I use two different data sources. The first is a new cross-country industry-level dataset that I use to describe the aggregate correlation between robot adoption and multinational production. The second is a detailed firm-level panel that I exploit to study how switching from domestic to foreign ownership changes firms' propensity to employ robots.

3.1 Industry-Level Data

I create a new cross-country industry-level dataset with information about MNEs' activities and industrial robots. Data about multinational activity come from the Analytical Multinational Enterprises Database (AMNE) of the Organization for Economic Cooperation and Development (OECD). For each country-industry-year tuple, the AMNE database contains a breakdown of gross output by domestic and foreign producers (i.e., MNEs) from 2005 to $2016.^{11}$

Data about industrial robots come from the International Federation of Robotics (IFR). The IFR aggregates cross-country firm-level information at the industry level to match the ISIC review 4 classification. The IFR data contain the number of deployed industrial robots at the country-industry-year level. These data are the most widely used source for robot adoption studies and are praised for their reliability (Graetz and Michaels, 2018; Acemoglu and Restrepo, 2020).

I complement the first two sources with standard industry accounts (e.g., total production, employment, wages, investment, fixed assets, exchange rate, and price deflators) from the Socio-Economic Account (SEA) database of the World Input-Output Database (WIOD). The final dataset contains information about 37 countries and 20 industries from 2005 to 2014. Industries include agriculture, mining, 15 two-digit manufacturing sectors, electricity and water supply, and construction. I express all nominal variables in USD million at constant prices. I provide more details about the sample creation and data cleaning in Appendix B.1. Table B.1 shows sample summary statistics.

3.2 Firm-Level Data

The ESEE Survey. Firm-level data come from the *Encuesta sobre Estrategias Empresariales* (ESEE, or Survey on Business Strategies) administered by the SEPI Foundation in Madrid.¹² The survey, spanning from 1990 to 2017, is a joint effort of the Ministry of Industry and the SEPI Foundation, and it is designed to be representative of the population of manufacturing firms with ten or more employees located in Spain. In 1990, the SEPI Foundation interviewed 2,188 firms divided into two categories. The first group contains firms with more than 200 employees. The second group is composed of a stratified sample of smaller firms employing 10-to-200 workers. From 1991 to 2017, the SEPI Foundation has surveyed about 1,800 firms each year and made an effort to minimize the sample deterioration due to either firms' exit or missing response.

Firms are assigned into 20 two-digit manufacturing industries that roughly match the NACE review 2 classification, and the survey contains a rich set of information about production process, sales, employment, cost and prices, technological adoption, and foreign trade. Crucially for my purposes, the ESEE data report firms' equity

¹¹The AMNE reports MNEs' production levels by country of origin. However, there is no distinction between vertical and horizontal FDI.

¹²Previous studies praise the reliability and accuracy of these data (Guadalupe et al., 2012; Doraszelski and Jaumandreu, 2013, 2018; Koch and Smolka, 2019; Stapleton and Webb, 2020; Koch et al., 2021).

composition and robot adoption choices. Following the International Monetary Fund (2007), I define a firm as foreign-owned if a company headquartered outside of Spain owns at least 10% of its capital. In terms of robot adoption, the survey asks firms "[...] whether their production process uses any of the following systems: 1. Computer-digital machine tools; 2. Robotics; 3. Computer-assisted design; 4. Combination of some of the above systems through a central computer (CAM, flexible manufacturing systems, etc.); 5. Local Area Network (LAN) in manufacturing activity". Based on the answer to this question, the SEPI Foundation constructs a binary indicator equal to 1 if a firm employs "Robotics" in a given year and 0 otherwise.¹³

Notwithstanding this richness, the ESEE data also come with some limitations. First, firms do not disclose the identity of their foreign owners.¹⁴ Second, the survey does not contain information about whether a firm is owned by a multinational company with headquarters in Spain. Finally, robot adoption information is only available at the extensive margin. The first limitation prevents analyzing otherwise interesting potential sources of heterogeneity. For example, one may conjecture that parents headquartered in places where robots are highly diffused are more likely to spur robot adoption than parents located elsewhere. Similarly, vertical and horizontal FDI may produce different effects. The second limitation prevents analyzing, for instance, if being acquired by a Spanish or foreign multinational delivers different robot adoption outcomes. Finally, the third limitation prevents studying the intensive margin of robot adoption and the effect of foreign ownership on different categories of industrial robots.¹⁵ Hopefully, more data will become available to inspect these differences in the future.

Data Cleaning and Sample Selection. I restrict my attention to the years between 1993 and 2014. I exclude years before 1993 to match the starting date of the IFR data. This choice also circumvents minor sample inconsistencies between the 1990 and subsequent survey waves. I exclude years after 2014 because robot adoption information is not available after that year.

I impose four sample selection criteria. First, I remove all firms that do not answer the survey for at least two consecutive years. I do so to avoid potentially non-random sample attrition. Second, I remove domestic firms that are involved in domestic mergers or divestitures during the sample period.¹⁶ Third, I remove Spanish firms that own

¹³Although industrial robots are not explicitly mentioned, Koch et al. (2021) show that robot adoption patterns in the ESEE data are consistent with the industry-level trends reported by the IFR.

¹⁴To be clear, if a firm is owned by a foreign company, the data only report the share of equity under foreign control.

¹⁵There are six types of industrial robots: articulated robots, SCARA (selective compliance assembly robot arm) robots, Cartesian robots, and parallel robots (International Federation of Robotics, 2019).

¹⁶The ESEE data contain explicit information about these changes. See Section 5.3, Other Types of Acquisitions for further details.

equity share of companies located abroad (i.e., perform "outward FDI").¹⁷ Fourth, I remove firms that are always foreign-owned or switch from domestic to foreign ownership multiple times. This criterion excludes greenfield FDI or firms already foreign-owned in 1993. Therefore, the final sample only includes always domestic-owned firms and firms that switch at most once from domestic to foreign ownership from 1994 onward.

One outstanding feature of the survey is that the robot adoption indicator is only available in eight years (1990, 1991, 1994, 1998, 2002, 2006, 2010, and 2014). Because the foreign ownership indicator has a yearly frequency, it is crucial to have the robot adoption indicator every year to study how switching ownership affects the likelihood of employing robots. I address this issue by filling robot adoption information forward between two consecutive non-missing survey answers.¹⁸ For instance, if a firm employs robots in 1994, I assume it also does it in 1995, 1996, and 1997. I update the indicator in 1998 based on the actual survey answer in that year and fill it forward until 2001 using the same criterion. I repeat this procedure in 2002, 2006, and 2010. For consistency, I impute the 1993 indicator with its value in 1991. Before imposing this criterion, about 65% of firm-year robot adoption information is missing. After, only about 8% is still missing. These are a few cases in which firms do not answer in two consecutive waves.

I also fill in missing values in the ownership structure and other ESEE variables. I interpolate missing values in the foreign ownership indicator using the same forward criterion I adopt for robot adoption. However, information about foreign ownership is available every year, and the share of missing observations in the sample is less than 1%. I also use the same rule to fill in missing values in other dichotomous indicator variables available every four years (e.g., other automation-type non-robotic technology adoption, manufacturing process, and exporter status). I replace missing values in the continuous variables (e.g., sales, employment, cost and prices, investment, physical inputs, etc.) with their average between two consecutive non-missing years, but I only do it if the length of the missing spell is shorter than three years.

Finally, I deflate nominal variables using firm-level or industry-level deflators. The ESEE provides a Paasche-type index that measures the average output price change between two consecutive periods for each active firm. It accommodates both mono and multi-product firms, and I use it to deflate firms' sales. I deflate physical inputs, R&D expenditure, wages, and investment using the one-digit manufacturing deflator from the *Instituto Nacional de Estadística* (INE). I use 2006 as the baseline year.

¹⁷The ESEE data report outward FDI activity only from 2000 onward. Hence, I can only apply this criterion as of that year. However, if a firm born in, say, 1997 starts doing outward FDI in, say, 2003, I exclude it from the sample.

¹⁸Forward imputation is consistent with the empirical finding that robot adoption is a lumpy investment (Humlum, 2021).

Sample Description. The final sample contains 3611 firms. Among them, 123 become foreign-owned at some point in time. Table A.1 shows summary statistics (mean and standard deviation) by ownership type using firms' lifetime characteristics. Therefore, for foreign-owned firms, I pool together pre and post-acquisition periods. Panel A reports summary statistics about the adoption of robots and other automation-type non-robotic technology (i.e., numerically controlled machines, CAD manufacturing, and flexible systems), whereas Panel B about the type of manufacturing production (i.e., batch, mass, continuous, and mixed manufacturing). Panels C and D contain information about firms' effort to acquire foreign technology and innovation activity, respectively. Finally, Panel E reports standard information about domestic and overseas activities (e.g., exports, number of export markets, wages, employees, investment, physical inputs, sales, and labor productivity). Table A.2 contains the description of each variable.

Although foreign-owned firms are only the 3% of the total, they outperform, on average, domestic-owned ones in each dimension. They are more productive, innovative, sell more, employ more workers, pay higher wages, and engage more in international trade than domestic firms indeed. However, because Table A.1 is constructed using firms' lifetime characteristics, it subsumes both foreign ownership selection and treatment effects. To inspect pre-acquisition differences, Figure A.1 shows the empirical probability density function (p.d.f.) of the log of employees, sales, physical inputs, R&D expenses, wages, and investment by ownership type. As in Table A.1, I estimate the empirical p.d.f. for domestic-owned firms based on their lifetime characteristics. However, unlike in Table A.1, I only estimate it for the years before the acquisition date for foreign-owned ones.

Overall, Figure A.1 illustrates that to-be-acquired firms outperform always domesticowned ones already before acquisition. This regularity is well-known in the literature (Guadalupe et al., 2012; Koch and Smolka, 2019; Bircan, 2019; Setzler and Tintelnot, 2021; Alfaro-Ureña et al., forthcoming). In Section 5, I will account for selection into foreign ownership to identify the effect of switching from domestic to foreign ownership on robot adoption.

4 Cross-Country Industry-Level Evidence

In what follows, I explore the relationship between multinational production (i.e., the gross output produced by foreign-owned firms) and robot adoption across countries and industries.

4.1 Multinational Production and Robot Adoption

The theoretical correlation between multinational production and robot adoption is ambiguous. Most FDI worldwide are vertical and, especially in the manufacturing sector, exploit production cost differences across countries (Antras and Helpman, 2004). As long as recipient countries are characterized by low wages, the robot adoption incentives are low in those places. From this viewpoint, one expects to find a negative correlation. However, foreign investors also care about the destination countries institutional quality, market size, technological level, and macroeconomic stability (Jardet, Jude and Chinn, 2022), which may also favor robots' diffusion. From this viewpoint, one expects to find a positive correlation.

The data favor the second hypothesis. Figure B.1 shows the average correlation between the log of the number of robots and the log of multinational production by country (across industries and years). The correlation is positive in most countries, the sample average is about 30%, and it is strongest in China, where it peaks at about 60%. Figure B.2 shows the linear fit of a regression of the log of the number of robots per thousand employees, which is a standard measure of robots' diffusion (Graetz and Michaels, 2018; Acemoglu and Restrepo, 2020), on the log of multinational production at the country-industry-year level. The estimated coefficient is about 0.5% and is significant at the 1% level. To further inspect the correlation between multinationals' activities and robot adoption, I estimate the following equation:

$$\log (R/N)_{cit} = \beta \log (MP)_{cit} + \mathbf{X}_{cit-1} \boldsymbol{\delta} + \eta_{cy} + \eta_{iy} + \varepsilon_{ciy}.$$
 (1)

 $\log (R/N)_{cit}$ is the log of the number of robots per thousand employees in country c's industry i in year t. $\log (MP)_{cit}$ is the log of gross output produced by foreign-owned firms in country c's industry i in year t. \mathbf{X}_{cit-1} is a matrix of time-varying country-industry level controls. It includes the one-year lag of the log of the capital-to-labor employment ratio and the interest rate-to-wage ratio. These variables control for country-industry level time-varying differences in relative production factor employment levels and prices, respectively. Finally, η_{cy} and η_{iy} capture any time-varying country and industry-level shocks potentially correlated with $\log (MP)_{cit}$. For instance, they account for the fact that countries or industries growing at different rates may experience heterogeneous patterns of robot adoption and multinationals' activity.

Table 1 reports the results. Column (1) shows the estimates without time-varying country-industry level controls, which I progressively add in Columns (2) and (3). The rightmost column contains the full specification and reports an estimate of about 0.28%, which is about half of the unconditional one in Figure B.2. As expected, Columns (2) and (3) show that robots' diffusion is more prominent where capital is more abundant

(i.e., the capital-to-employee ratio is higher) and cheaper (i.e., the interest rate-to-wage ratio is lower) than labor.

Dependent Variable:	Log(Robot (1)	ts/Thousand E (2)	$\begin{array}{c} \text{mployees} \\ (3) \end{array}$
$Log(Multinational Production)_{cit}$	0.28^{***} (0.03)	0.27^{***} (0.03)	0.28^{***} (0.03)
$Log(Capital Stock/Thousand Employees)_{cit-1}$	(0.00)	0.13^{***}	0.17^{***}
$Log(Interest Rate/Wages)_{cit-1}$		(0.04)	-0.10^{**}
Industry-Year FE Country-Year FE	Yes Yes	Yes Yes	(0.04) Yes Yes
Observations Estimator	6,563 OLS	5,870 OLS	5,744 OLS

Table 1: CROSS-COUNTRY INDUSTRY-LEVEL CORRELATIONS

Note: The unit of observation is a country-industry-year tuple. $Log(Robots/Thousand Employees)_{cit}$ is the log of the number of robots per thousand employees in country c's industry i in year t. $Log(Multinational Production)_{cit}$ is the log of gross output produced by foreign-owned firms in country c's industry i in year t. $Log(Capital Stock/Thousand Employees)_{cit}$ is the capital stock per thousand employees in country c's industry i in year t. $Log(Interest Rate/Wages)_{cit}$ is the ratio between the capital interest rate and gross wages in country c's industry i in year t. Heteroscedasticity-robust standard errors in parenthesis. Significance levels: *** 0.01, ** 0.05, * 0.1.

Table B.2 shows that the baseline results are robust to omitting the top five robotized countries in the sample (i.e., China, Germany, Japan, South Korea, and the United States) and the automotive industry, which accounts for over 80% of the total worldwide robot stock in any sample year. This result strengthens the confidence about the robustness of the correlation in Table 1. Finally, in Appendix B.3.1 I show that a qualitatively similar pattern also holds using a long-difference specification, which builds confidence that the baseline results do not merely pick up pre-sample trends.¹⁹

5 The Foreign Ownership Robot Adoption Premium

The cross-country evidence highlights a positive and robust correlation between multinational production and robot adoption. However, reverse causality and omitted variable bias prevent a causal interpretation. To make further progress in understanding whether becoming foreign-owned affects the probability of robot adoption, I exploit

¹⁹In an unreported specification, I re-estimated equation (1) using only multinational production from the top five robotized countries in the sample. The idea was to check if FDI coming from highly robotized countries correlated more strongly with robot adoption in the destination. However, I found no evidence in favor of this hypothesis.

the ESEE firm-level data. First, I provide descriptive evidence about the relationship between foreign ownership and robot adoption at the firm-level. Second, I deal with selection and estimate how switching to foreign ownership increases the probability of employing robots (i.e., the foreign ownership robot adoption premium). Third, I evaluate the explanatory power of different mechanisms consistent with the results.

5.1 Foreign Ownership and Robot Adoption: A First Look

Figure A.2 shows how robot adoption has evolved by ownership type. Although the proportion of robot adopters has nearly doubled in both sub-samples over time, the share of robot adopters in any given year among foreign-owned firms is about twice as large than among domestic-owned ones. Figure A.3 shows that a similar pattern applies in the cross-section across industries. The fact that foreign-owned firms are systematically more likely to employ robots than domestic ones is new in the literature.

To move towards quantifying the foreign ownership robot adoption premium, I estimate the following event-study specification:

$$R_{f(j)t} = \sum_{s=-\underline{s}}^{\overline{s}} \theta_s O_{ft}^s + \mathbf{X}_{ft-1} \boldsymbol{\beta} + \alpha_f + \alpha_{jt} + \varepsilon_{f(j)t}.$$
 (2)

 $R_{f(j)t}$ is a dummy equal to one if firm f in industry j employs a robot in year t and zero otherwise. The event-time dummy is $O_{ft}^s := \mathbf{1}\{t = \tau_f + s\}$ for $s \in \{-\underline{s}, \ldots, \overline{s}\}$. $\mathbf{1}\{\ldots\}$ is an indicator function, and τ_f is the year in which firm f is acquired. Therefore, O_{ft}^s is a dummy identifying the periods before or after the foreign acquisition date. I saturate the model by defining $O_{ft}^{\overline{s}} := \mathbf{1}\{t > \tau_f + \overline{s}\}$ and $O_{ft}^{\underline{s}} := \mathbf{1}\{t < \tau_f - \underline{s}\}$. I set $\underline{s} = \overline{s} = 5$. I also normalize $\theta_{-1} = 0$. \mathbf{X}_{ft-1} is matrix of firm-year level controls. It includes one, two, and three-year lags of (the log of) firm sales and the one-year lag of sales' growth rate. These controls account for observed time-varying differences in size and performance among firms. Finally, α_f and α_{jt} denote firm, respectively industry-year, fixed effects.

I estimate equation (2) only using the sub-sample of the 123 ownership switchers. The event-study coefficients θ_s are identified using within-firm variation after controlling for firm-level time-varying observed differences and industry-level (possibly timevarying) common shocks to all firms. The identifying assumption is that, after including such controls, not-yet-acquired firms are a credible counterfactual for acquired ones.²⁰ Figure 1 shows the results. As of the first year after the acquisition date, new affiliates

 $^{^{20}}$ I also estimate equation (2) including never acquired firms in the control group. In this case, the identifying assumption is that, after including the controls on the right-hand side of equation (2), both not-yet-acquired and never acquired firms constitute a credible counterfactual for acquired ones. Figure C.1 shows the results, which are qualitatively similar to the baseline.

become about 35%, on average, more likely to employ robots than in the year before the acquisition. There are no statistically significant pre-trends at any conventional statistical level.



Figure 1: EVENT-STUDY

Note: The Figure plots the estimates (and the 95% confidence intervals around them) that I obtain from equation (2). The unit of observation is a firm-year pair, and I cluster standard errors by firm. I restrict the sample to the subset of acquired firms. Starting from the first year after the acquisition date, new affiliates become about 35% more likely to employ robots than in the year before the acquisition.

Several recent papers have shown that estimating event-study specifications like equation (2) with a two-way fixed-effects estimator may fails to recover the treatment effect of interest even if assignment into treatment is random.²¹ This happens when the treatment roll-out is staggered and the treatment effects evolve over time. Intuitively, the inclusion of already treated units in the control group for some cohorts gives rise to a "forbidden comparison" (Borusyak et al., 2021), and the two-way fixed effects estimator delivers variance-weighted averages of heterogeneous treatment effects which are hard to interpret.

In my setting, this concern is warranted if the probability of robot adoption varies over time depending on the acquisition year. To deal with this issue, I use the method proposed by Sun and Abraham (2021) and estimate cohort-specific dynamic treatment effects, which I then aggregate using the size of each cohort as weight. Figure C.2 shows

²¹See, e.g., De Chaisemartin and d'Haultfoeuille (2020), Sun and Abraham (2021), Borusyak et al. (2021), Callaway and Sant'Anna (2021), and Goodman-Bacon (2021).

the results. Overall, the two figures deliver a qualitatively similar pattern and, except for the first year post-acquisition, statistically equal estimates at the 5% level.

5.2 The Effect of Foreign Ownership on Robot Adoption

Empirical Strategy. Although suggestive of a causal relationship, the estimates in Figure 1 may be contaminated by selection effects. In particular, one may worry that unobserved firm-level transitory shocks that correlate with foreign acquisition may also drive robot adoption (Blundell and Costa Dias, 2009). For instance, some firms may foresee the foreign acquisition and, therefore, restructure their operations and investment (in robots or other complementary technologies) already before the event. Whereas the absence of pre-trends in Figure 1 holds back this hypothesis, I explicitly deal with this concern in what follows.

The ideal way to address this issue would be to randomly assign foreign ownership to domestic firms and see how their robot adoption choices change, if at all. Unfortunately, this approach is unfeasible. To make progress, I use an inverse propensity score reweighing (IPRW) estimator à la Guadalupe et al. (2012).²² This procedure combines a difference-in-differences approach with a propensity score reweighing procedure, and compares changes in robot adoption of acquired firms with those of similar firms that remain under domestic control.

The estimator works in two stages. In the first stage, I estimate the probability that each firm is acquired by a foreign company based on an exhaustive set of firms' observable characteristics in levels and growth rates.²³ To do so, from 1994 to 2014, I include firms acquired in that year in the treatment group and firms that are always domestic-owned in the control group. Then, I stack together treated and control firms across all years and estimate, for each firm, the probability of becoming foreign-owned as a function of the following variable: sales' growth rate, log sales, log number of employees, log wages, log investment, log physical inputs, log R&D expenses, log exports, and number of export markets.²⁴ All variables refer to the one-year lag before the acquisition date, and I call them "selection controls". Because of missing values, I can

 $^{^{22}}$ A number of papers building upon Guadalupe et al. (2012) also employ this estimator (Koch and Smolka, 2019; Koch et al., 2021). My implementation of the estimator closely follows the standard practice in this literature. DiNardo, Fortin and Lemieux (1996) show that the IPRW estimator has better finite sample properties than other matching techniques such as propensity score matching.

²³The purpose to account for all the relevant time-varying observed determinants that predict foreign acquisition at the firm level (Guadalupe et al., 2012; Koch and Smolka, 2019).

²⁴Table C.1 shows first-stage probit regressions adding one variable at a time. The rightmost column of the Table highlights that firms are more likely to become foreign-owned if they are growing faster than average (i.e., with a higher growth rate of sales between the acquisition year and the one before), are larger (i.e., employ more employees than average), more innovative (i.e., spend more on R&D than average), and export more.

successfully compute the propensity score only for 79 out of 123 total foreign-owned firms.²⁵ As Guadalupe et al. (2012), I also exclude from the control group firms that do not fall within the common support. Then, I assign a weight of $1/\hat{p}$ to acquired firms and of $1/(1-\hat{p})$ to non-acquired one, being \hat{p} the estimated first-stage probability (i.e., the propensity score). This reweighing scheme aims at making the two groups of firms comparable on observables before the acquisition.²⁶

In the second stage, I estimate the following equation on the reweighed sample:

$$R_{f(j)t} = \theta O_{ft} + \alpha_f + \alpha_{jt} + \varepsilon_{f(j)t}.$$
(3)

Notation follows from equation (2). The main differences with equation (2) are that I suppress time-varying firm-level controls (\mathbf{X}_{ft-1}) and estimate a single value for the foreign ownership robot adoption premium (θ). The identifying assumption is that, post-reweighing and after controlling for the included fixed effects, treatment is random. Under this assumption, the IPRW estimator identifies the effect of becoming foreignowned on the probability of robot adoption (i.e., the foreign ownership robot adoption premium).

Results. Table 2 reports the results of estimating equation (3). For comparison, Columns (1) and (2) show the OLS estimates without, respectively with, firm and industry-year fixed effects. Column (1) shows an unconditional estimate for θ of about 28%, whereas accounting for firm fixed effects and industry trends delivers an estimate of about 11%. Column (3) shows the results when including all the selection controls as additional regressors. In this case, the estimated average coefficient is about 12%. Column (4) shows the results using the IPRW estimator, which I consider the preferred specification. The estimates reveal that, post-acquisition, affiliates are about 10% more likely to employ robots than comparable domestic producers.²⁷ Although the estimates in Columns (2), (3), and (4) are not statistically different from each other, the OLS point estimates are slightly upward-biased, consistently with positive self-selection into foreign ownership.

 $^{^{25}}$ As a robustness check, I also re-estimated the propensity score only as a function of lagged log sales and lagged log number of employees. In this case, I can estimate it for all firms in the treatment group. Using this procedure, I estimate an average premium of 9%, which is statistically indistinguishable from that in the rightmost column of Table 2.

²⁶Table C.2 shows that, before reweighing, I always reject the hypothesis that the treatment and control group have the same average characteristics at any conventional statistical level. After reweighing, I never reject it at the 1% significance level.

 $^{^{27}}$ For benchmarking, the average probability that a firm employs robots in the sample is about 17%. The estimated premium about 58% of it.

Dependent Variable:	$\operatorname{Robot}_{ft}$			
	(1)	(2)	(3)	(4)
Foreign-Owned _{ft}	0.28***	0.11^{*}	0.12^{*}	0.10**
	(0.05)	(0.06)	(0.07)	(0.05)
Firm FE	No	Yes	Yes	Yes
Industry-Year FE	No	Yes	Yes	Yes
Selection Controls	No	No	Yes	No
Observations	21,010	21,010	19,561	18,520
Estimator	OLS	OLS	OLS	IPRW

Table 2: FIRM-LEVEL BASELINE REGRESSIONS

Note: The unit of observation is a firm-year pair. Robot_{ft} is a binary variable equal to 1 if firm f employs a robot in year t and 0 otherwise. Foreign-Owned_{ft} is a binary variable equal to 1 if firm f is foreign-owned in year t and 0 otherwise. Selection controls include the one-year lag of sales' growth rate, log sales, log number of employees, log wages, log investment, log physical inputs, log R&D expenses, log exports, and number of export markets. In Column (4), I reweigh acquired firms by $1/\hat{p}$ and non-acquired ones by $1/(1-\hat{p})$, being \hat{p} the estimated probability of becoming foreign-owned. Cluster standard errors by firm in parenthesis. Significance levels: *** 0.01, ** 0.05, * 0.1.

Finally, Figure C.3 plots the dynamic estimates that I obtain by estimating equation (2) on the reweighed sample. The main difference with the baseline specification is that I drop the selection controls on the right-hand side of the equation and apply the inverse probability reweighing scheme instead. Whereas Figure C.3 delivers similar post-acquisition patterns as Figure 1, the reweighing procedure further assuages pre-acquisition anticipation effects. This result strengthens the interpretation of the premium and clarifies the role of foreign ownership in spurring robot adoption. Despite the fact that the adoption of robots is a growing phenomenon, MNEs accelerate and spread it.

Heterogeneity. Even if all firms experienced the same ownership shock, they might respond differently depending on their specific activity and surrounding environment. In what follows, I investigate if the premium depends on the degree of robots' diffusion in a given industry. This heterogeneity is relevant to understand if some sectors may be more exposed to the outcomes of multinational-induced robot adoption, which I discuss in Section 6.

I proxy industry-level robot adoption as follows. First, I compute the number of industrial robots deployed in each Spanish manufacturing industry in 1993. I choose this year because it is the first available one in the IFR data and to prevent the measure from being affected by robot adoption from 1993 onward.²⁸ Second, I group industries

²⁸I follow Koch et al. (2021)'s cross-walk tables to match IFR and ESEE industry codes.

into terciles.²⁹ The first tercile contains industries characterized by a low initial robot adoption stock (e.g., manufacture of coke and pharmaceutical), whereas the last tercile contains industries characterized by a high initial robot adoption stock (e.g., automotive and electronics). The second one includes those in the middle (e.g., textiles).

Column (1) of Table C.3 reports the results of equation (3) when interacting the foreign-ownership binary indicator with my industry-level proxy of robot adoption.³⁰ The estimates reveal that the premium is about 27% for firms in the first tercile. This figure is about three times larger than the baseline estimate. Firms in the second tercile report a premium of 13% instead, which is statistically indistinguishable from the baseline one. Finally, I do not reject the hypothesis that the premium for firms in the last tercile is nil. Columns (2), (3), and (4) show the results when estimating equation (3) on each sub-sample separately. The results are consistent with Column (1). Overall, the estimates suggests that initially less robotized sectors are more exposed to multinational-induced robotization and, potentially, to its implications.

5.3 Why Do Foreign Ownership Boosts Robot Adoption?

Alternative Hypotheses. Different mechanisms may explain why switching from domestic to foreign ownership boosts robot adoption. For instance, previous literature using the ESEE data shows that the possibility of expanding the customer base into foreign markets through the network of the parent is a central driver of product and process innovation (Guadalupe et al., 2012) and skill upgrading (Koch and Smolka, 2019). To the extent that robot adoption is complementary to these activities and allows to increase production and expand in global markets, the foreign market access channel is also relevant in my context.

An alternative hypothesis is that foreign acquisitions alleviate credit constraints, thus allowing firms to move closer to their first-best investment level (Harrison and McMillan, 2003; Manova et al., 2015). Another possible explanation is that foreign parents directly transfer knowledge about robot adoption (or other complementary inputs) to their affiliates. Indeed, technology flows within multinational networks are extensively documented in the literature (Branstetter et al., 2006; Keller and Yeaple, 2013; Bilir and Morales, 2020).

In what follows, I exploit the richness of the ESEE survey to evaluate the importance

²⁹The first tercile includes 211 firms (11 of which become foreign-owned), the second one includes 1472 firms (35 of which become foreign-owned), and the third one includes 717 firms (31 of which become foreign-owned). I choose terciles to ensure there are enough foreign-owned firms in each bin. However, I find qualitatively similar results when using, e.g., quartiles.

³⁰The inclusion of industry-trends in equation (3) accounts for any other industry-level, possibly time-varying, characteristics (e.g., share of tasks that can be performed by robots) that may also affect the size of the premium.

of these three mechanisms. I proceed in two steps. First, I test whether switching to foreign ownership is associated with increased foreign market access, reduced credit constraints, and technology transfers. Second, I evaluate the explanatory power of these three channels for robot adoption.

Testing Mechanisms. To study whether switching from domestic to foreign ownership grants access to foreign markets, credit, or superior technology, I estimate again equation (3) but replace the robot indicator on the left-hand side with variables related to the three mechanisms I evaluate.

To measure whether a firm has access to foreign markets via its foreign parent network, I exploit a question in the ESEE survey that asks firms by what means they access export markets. The possible answers are that they export via their foreign parents (either using their distribution channel or directly selling to them), their own means, specialized intermediaries, collective actions, or other means. Based on the answer to this question, the SEPI Foundation constructs a binary indicator equal to 1 if a firm exports via its foreign parent in a given year and 0 otherwise. The variable is only defined for exporters and, by construction, is always zero for domestic-owned firms. I use this variable as a proxy for foreign market access.

To evaluate if credit constraints prevent robot investment, I test whether acquired firms increase their external R&D expenditures per worker, an activity typically subject to credit constraints (Brown et al., 2012).³¹ Finally, although thoroughly testing for technology transfers would require (unfortunately unavailable) data about intra-firm activities, the ESEE reports firm expenditure on licenses and technical aid from abroad, which I use to proxy technology transfers. In particular, I define a variable equal to 1 if a firm imports foreign technology in a given year and 0 otherwise. Then, I check if acquired firms are more likely to import technology from abroad, possibly from their parents, than domestic firms.

Table 3 shows the results. Column (1) shows that, conditional on exporting, acquired firms are about 44% more likely to do it via their foreign parental network than any other means. By contrast, they do not increase their external R&D per employee (Column 2) and are not more likely to import foreign technology (Column 3), which holds back the credit constraints and technology transfers hypotheses.

To provide further evidence about affiliates' expansion into global markets following the acquisition date, Figure C.4 shows the event-study results that I obtain by estimating equation (2) on the reweighed sample and using Exp. via Foreign Parent_{ft} as outcome variable. Consistently with Table 3, there is a sharp increase in the probability of exporting via the foreign parental network in the first four years after the acquisition

³¹I obtain similar results if I use, e.g., external R&D expenditures over total investment or sales.

Dependent Variables:	Exp. via Foreign $\operatorname{Parent}_{ft}$ (1)	$\begin{array}{c} \text{Log}(\text{Ext. R\&D/Employees})_{ft} \\ (2) \end{array}$	Imp. of Foreign Tech. _{ft} (3)
$\operatorname{Foreign-Owned}_{ft}$	0.44^{***}	0.01	0.03
Firm FE	(0.07) Yes	(0.39) Yes	(0.02) Yes
Industry-Year FE	Yes	Yes	Yes
Observations Estimator	3,084 IPRW	20,510 IPRW	20,526 IPRW

Table 3: FIRM-LEVEL MECHANISM REGRESSIONS (STEP ONE)

Note: The unit of observation is a firm-year pair. Exp. via Foreign $\operatorname{Parent}_{ft}$ is binary variable equal to 1 if firm f exports via its foreign parental network at time t and 0 otherwise. $\operatorname{Log}(\operatorname{Ext. R\&D}/\operatorname{Employees})_{ft}$ is the log of one plus the expenditure on external R&D per employee. Hence, it accounts both for the intensive and extensive margin of external R&D. Imp. of Foreign Tech. $_{ft}$ is binary variable equal to 1 if firm f imports licenses and technical aid from abroad at time t and 0 otherwise. Foreign-Owned $_{ft}$ is a binary variable equal to 1 if firm f is foreign-owned in year t and 0 otherwise. In all columns, I reweigh acquired firms by $1/\hat{p}$ and non-acquired ones by $1/(1 - \hat{p})$, being \hat{p} the estimated probability of becoming foreign-owned. Cluster standard errors by firm in parenthesis. Significance levels: *** 0.01, ** 0.05, * 0.1.

date. I also estimate the same specification using the log of exports as outcome variable. Figure C.5 shows that, post-acquisition, affiliates increase by about 30% their exports compared to the year before the acquisition. Hence, the possibility of exporting via the foreign parental network is also associated with an increase in the overall volume of exports. There are no pre-trends at the 5% level in both cases.

Next, I evaluate which of these three mechanisms, if any, has the largest explanatory power for robot adoption. To do so, I estimate again equation (3) but replace the binary indicator for foreign ownership on the right-hand side with the three proxies for the mechanisms of interest, which I add progressively.

Table 4 shows the results. Column (1) shows that firms that access export markets via their foreign parental network are about 17% more likely to employ robots than those that export via any other channel.³² Columns (2) and (3) show that this estimate is robust to including my proxies for credit constraints and import of foreign technology, which do not significantly correlate with robot adoption.

To summarize, the results point out to the following mechanism. Upon acquisition, affiliates have the chance of expanding their customer base abroad thanks to their foreign parental network. However, firms need to scale-up production to translate higher potential demand into actual sales, and robot adoption is one way to achieve this result. This results is also consistent with previous literature showing that the ability to serve foreign markets is a crucial driver of technology adoption and innovation (Lileeva and Trefler, 2010; Bustos, 2011).

 $^{^{32}}$ To be clear, the treatment group includes acquired firms exporting via their foreign parent, whereas the control group contains domestic-owned exporters and acquired firms that do not export via their foreign parent.

Dependent Variable:		$\operatorname{Robot}_{ft}$	
	(1)	(2)	(3)
Exp. via Foreign $\operatorname{Parent}_{ft}$	0.17**	0.18**	0.17**
·	(0.08)	(0.08)	(0.08)
$Log(Ext. R\&D/Employees)_{ft}$		-0.002	-0.002
		(0.005)	(0.005)
Imp. of Foreign Tech. $_{tt}$			0.09
			(0.08)
Firm FE	Yes	Yes	Yes
Industry-Year FE	Yes	Yes	Yes
Observations	3,080	3,074	3,074
Estimator	IPRW	IPRW	IPRW

Table 4: FIRM-LEVEL MECHANISM REGRESSIONS (STEP TWO)

Note: The unit of observation is a firm-year pair. Exp. via Foreign Parent_{ft} is binary variable equal to 1 if firm f exports via its foreign parental network at time t and 0 otherwise. Log(Ext. R&D/Employees)_{ft} is the log of one plus the expenditure on external R&D per employee. Hence, it accounts both for the intensive and extensive margin of external R&D. Imp. of Foreign Tech._{ft} is binary variable equal to 1 if firm f imports licenses and technical aid from abroad at time t and 0 otherwise. Foreign-Owned_{ft} is a binary variable equal to 1 if firm f is foreign-owned in year t and 0 otherwise. In all columns, I reweigh acquired firms by $1/\hat{p}$ and non-acquired ones by $1/(1-\hat{p})$, being \hat{p} the estimated probability of becoming foreign-owned. Cluster standard errors by firm in parenthesis. Significance levels: *** 0.01, ** 0.05, * 0.1.

The Heterogenous Effects of Increased Market Access. Table C.4 digs deeper into the role of exporting via the foreign parent. In particular, I run again the specification in Column (1) of Table 4 but interact Exp. via Foreign $Parent_{ft}$ with binary indicators of previous ownership (Column 1), export status (Column 2), and both previous ownership and export status (Column 3).

Column (1) shows that firms that access export markets via their foreign parental network in the same year that they are acquired are about 30% more likely to employ robots than acquired firms that export via other means.³³ This figure is 8 p.p. lower for firms that access export markets via their foreign parent in the current period but were already foreign-owned in the previous one.³⁴ Consistently with Table 2, being foreign-owned in the previous year also correlates with robot adoption in the current one.

Column (2) shows that exporting via the foreign parental network increases by

³³Because only acquired firms can export via their foreign parental network, Exp. via Foreign Parent_{ft} = 1 and Foreign-Owned_{ft-1} = 0 denotes firms acquired in t that start exporting via the foreign parental network in that year.

³⁴For this group, the estimated probability is 0.30 + 0.29 - 0.37 = 22%.

about 45% the probability of robot adoption for firms that did not export at all in the previous period. By contrast, this probability is about 30% for firms that were already exporting in the previous year.³⁵ Consistently with Koch et al. (2021), I also find that being an exporter positively correlates with the probability of robot adoption.

Finally, Column (3) shows that exporting via the foreign parents has the largest impact on robot adoption for firms that were not foreign-owned and did not export in the previous period. For them, the estimated probability of adoption is 46%. This probability is about 33% for firms that export via their foreign parent in the current period and were already foreign-owned but did not export in the previous one. A similar figure also applies to already exporting firms.³⁶

Other Types of Acquisitions. One interesting question is whether different types of acquisitions, e.g., from a Spanish multinational or domestic firm, produce different robot adoption outcomes. Answering this question is relevant because it reveals whether policies promoting foreign investment may deliver different outcomes than those promoting, for instance, the activities of multinationals headquartered in the home country or domestic mergers. In principle, given the documented importance of increased market access through the parental network, I expect that being acquired by a Spanish multinational produces similar outcomes. Unfortunately, as already mentioned in Section 3.2, the ESEE survey does not report this information, which makes it unfeasible to evaluate this hypothesis.

However, the data do contain information about domestic mergers. In particular, each firm reports whether it changed legal status in a given year. There are six possible cases: (1) no change, (2) it has split, (3) it has acquired other firms, (4) it has born after a split process, (5) it is the result of a domestic merger process, or (6) it has changed the trademark and legal form. As in the baseline exercise, I include firms that never experience any change in the control group. However, I now include in the treatment group firms that are the result of a domestic merger process. There are 33 firms that meet this criterion.³⁷ Even if the size of the treatment group is limited, the ESEE data allow to follow these firms over time. Therefore, I can create a binary indicator equal to 1 if a firm was acquired by another Spanish company (i.e., it is the result of a domestic merger process) in year $s \leq t$ and 0 otherwise.

I estimate again equation (2) using this new sample. The only difference is that the O_{ft}^s dummy(-ies) identifies the periods before or after the domestic acquisition event.

³⁵For this group, the estimated probability is 0.45 + 0.14 - 0.29 = 30%.

³⁶For the first group, the estimated probability is 0.46 + 0.28 - 0.40 = 33%. For the second, it is 0.46 + 0.28 - 0.40 + 0.13 - 0.12 = 35%. However, these two estimates are not statistically different.

 $^{^{37}}$ As for future foreign-owned firms, Figure C.6 shows that to-be-acquired firms by a domestic company are a positively selected sub-sample of the population of Spanish manufacturing firms.

Figure C.7 shows that, post-acquisition, firms acquired by a domestic parent are not significantly more likely to employ a robot than in the year before the acquisition. To deal with similar identification concerns as in Section 5.2, I re-adapt and use again the IPRW estimator I described for equation (3).³⁸ Table C.6 shows the results. Whereas the OLS estimates deliver a statistically positive correlation between being acquired by a domestic company and robot adoption, the estimate becomes statistically indistinguishable from zero after controlling for selection using the IPRW estimator.

The previous discussion about the importance of foreign market access offers a potential explanation for this result. By definition, non-multinational Spanish parents do not provide their affiliates with access to global markets through their multinational network. Hence, domestic mergers may not grant a sufficiently higher demand level to justify paying the fixed cost of robot adoption.

5.4 Other Outcomes

Robots are rather expensive and sophisticated machines to use.³⁹ If foreign owners boost their adoption, it is plausible that they also restructure firms' production process more broadly. For instance, using a similar specification as in equation (3), Table C.7 shows that newly acquired firms become about 8% less likely to perform batch manufacturing, i.e., low-scale serial production, and about 6% more likely to engage in continuous manufacturing, i.e., 24/7 high-scale production activity. Although this type of production naturally lends itself to robot deployment, foreign ownership my also bring about other changes. In what follows, I investigate whether switching to foreign ownership boosts the adoption of other automation technology and, more in general, how firms' performance changes after the acquisition.

Other Automation Technology. One interesting question is whether firms acquired by a foreign company also become more likely to employ other automation technology beyond robots and, if so, which ones. Answering this question is useful to understand the nature of the technological plan that foreign parents have for their affiliates in Spain.

³⁹Acemoglu and Restrepo (2020) document that this cost ranges from about 44,000\$ per robot to about 88,000\$. These amounts are about 2% and 4% of the median annual sales value in the sample.

³⁸In the first step, I estimate the probability of being acquired by another domestic firm as a function of the lagged (log of) number of employees, (log of) sales, (log of) wages, (log of) physical inputs, exporter status, sales and employment growth rate. Table C.5 shows that, after reweighing, I do not reject the hypothesis that the treatment and control group have the same average characteristics at the 1% significance level. Notice that the selection controls differ from the baseline exercise. I change them to achieve a better in-sample covariates' balance between groups than I would get using the same set of the main exercise. As a robustness check, I also compute the propensity score using the same selection controls as in the baseline exercise. Reassuringly, I obtain similar second-stage results.

Thanks to the richness of the ESEE survey, I can use the same survey question that reports if a firm uses a robot (see Section 3.2) to construct binary indicators for whether firms employ computer-assisted design (CAD) manufacturing (i.e., a technology that allows doing computerized process design), numerically controlled machines (i.e., automatic machines that perform specific routine tasks), flexible design systems (i.e., computerized machines that help to connect robotic, CAD, and numerically controlled machines together), or any of those technologies. Whereas CAD manufacturing and flexible systems exhibit some complementarity with robots, numerically controlled machines should be understood as imperfect substitutes.⁴⁰ More in general, all these technologies are older, have lower labor-replacing potential, and scale-up firms' output less than robots do (Acemoglu and Restrepo, 2020).

I estimate equation (3) again but replace the robot adoption indicator with one of those new variables at a time. Table C.8 shows the results. Column (1) indicates that newly acquired firms become about 5% more likely to employ CAD manufacturing. However, Columns (2) and (3) show that they do not also become more likely to use numerically controlled machines or flexible systems. Finally, Column (4) shows that foreign acquisitions do not significantly spur the joint adoption of any of those automation-type non-robotic technologies.

Overall, these findings suggest that foreign parents deem robots to be different. Rather than investing in any automation-driven technology type, newly acquired firms only acquire the most advanced automation technology that allows them to scale-up operations more significantly. Whereas they may also decide to adopt complementary technology, there is no evidence that they invest in lower-quality substitutes.

Gains and Distributional Outcomes. The results documented so far suggest that foreign ownership generates substantial gains for acquired firms (e.g., enabling production scale-up to expand in the global market). An interesting question is how conspicuous are these gains and whether they are evenly shared within the firm.

To answer these questions, I estimate again equation (3) but use several measures of firm performance as outcome variables. Table C.9 reports the results.⁴¹ The first four columns reveal that switching to foreign ownership induces sizable firm-level gains. Column (1) shows that, upon acquisition, firms increase sales by by about 37%, whereas Columns (2) and (3) show that they also increase employment and wages by 22%, respectively 9%.⁴² Labor productivity grows by about 12% after the acquisition (Column

 $^{^{40}}$ The main advantage of robots is that they can be reprogrammed to perform multiple tasks. By contrast, numerically controlled machines can usually perform a single task only.

⁴¹Notice that the outcomes are expressed in logs. Therefore, the average effect must be computed as $100 \times (\exp(x) - 1)\%$, being x the estimated coefficient.

⁴²This wage premium is consistent with the estimates of Setzler and Tintelnot (2021) for the US.

4). However, Column (5) suggests that foreign acquisition does not evenly benefits all production factors. The ratio between sales and wages increases by about 23% indeed, so the labor share in revenues fall and firms tilt production towards capital-intensive activities.⁴³

Overall, foreign acquisitions create both gains and distributional outcomes within firms. One interesting question is how much of these changes are due to robot adoption versus other forces. I address this question and discuss the industry-level implications of FDI and robot adoption in the next section.

6 The Implications of MNE-Induced Robotization

The previous results show that foreign-owned firms are more likely to deploy robotic technology. Whereas robot adoption boosts firm activities, it also redistributes income away from labor within firms. I leverage these insights to introduce a dynamic partial equilibrium model of firm investment based on Doraszelski and Jaumandreu (2018) and Humlum (2021), which I use to further investigate the aggregate gains and distributional consequences of multinational-induced robotization in the Spanish manufacturing industry during the sample period.

6.1 A Model of Foreign Ownership and Robot Adoption

Set Up. Firms employ physical inputs and labor with constant elasticity of substitution to produce final output. Their production function features labor-augmenting and total factor productivity.⁴⁴ Firms can be either domestic or foreign-owned. As in the previous section, foreign ownership is irreversible. If firm f is foreign-owned in year t, it will also stay under foreign ownership in any $\tau > t$. By contrast, domestic-owned firms in year t become foreign-owned with finite probability in the next period.⁴⁵

Firms can invest in robots to maximize net (of the fixed robot adoption cost) profits (π_{ft}) . I assume that current ownership and robot adoption choices affect future laboraugmenting and total factor productivity, which evolve according to an endogenous

⁴³This result is consistent with a large body of past literature showing that foreign-owned firms are more capital intensive than domestic-owned ones (Koch and Smolka, 2019; Sun, 2020).

⁴⁴I make this modeling choice following Doraszelski and Jaumandreu (2018), who show that the observed employment and price dynamics in Spanish manufacturing are more consistent with laboraugmenting rather than capital-augmenting technological change. However, the main insights of this section hold no matter the type of factor-biased productivity assumed.

⁴⁵This choice is dictated by parents' anonymity in the ESEE data. Guadalupe et al. (2012) model foreign acquisition in the same way.

Markov process.⁴⁶ Let $\Omega_{ft} = \{w_{ft}, r_{ft}, B_{ft}, z_{ft}, O_{ft-1}, R_{ft-1}\}$ be firm f's state vector in year t. It includes current labor (w_{ft}) and physical inputs (r_{ft}) input prices, laboraugmenting (B_{ft}) and total factor (z_{ft}) productivity, past ownership structure (O_{ft-1}) and robot adoption choices (R_{ft-1}) . Firms choose physical inputs (M_{ft}) , labor (L_{ft}) , and invest in robot adoption (R_{ft}) .⁴⁷ Firm f's problem in year t reads:

$$V_t(\Omega_{ft}) = \sup_{\{M_{ft} \ge 0, R_{ft} \in \{0,1\}\}} \{\pi(\Omega_{ft}) + \beta \mathbb{E}_t [V_{t+1}(\Omega_{ft+1})]\}$$
(4)

s.t.
$$y_{ft} = z_{ft} \left(M_{ft}^{\frac{\sigma-1}{\sigma}} + B_{ft}^{\frac{1}{\sigma}} L_{ft}^{\frac{\sigma-1}{\sigma}} \right)^{\frac{\sigma}{\sigma-1}}, \quad \sigma > 0, \ \sigma \neq 1$$
 (5)

$$\log(B_{ft}) = \beta \left(\log(B_{ft-1}), R_{ft-1}, O_{ft-1} \right) + u_{ft}$$
(6)

$$\log(z_{ft}) = \zeta \left(\log(z_{ft-1}), R_{ft-1}, O_{ft-1} \right) + \xi_{ft}$$
(7)

The expectation in equation (4) operates over future productivity levels and ownership structure. The production function in equation (5) contemplates that physical inputs and labor are either gross complements, $\sigma \in (0, 1)$, or gross substitutes, $\sigma > 1.^{48}$ Equation (5) also implies a different role of total factor (Hicks-neutral) and laboraugmenting productivity. While z_{ft} uniformly shifts output, B_{ft} changes the relative productivity of physical inputs and labor within the firm.

Equations (6) and (7) state that current productivity depends on its past value, ownership structure, and robot adoption choices. Moreover, both productivity types depend on a expected term (i.e., $\beta(...)$ and $\zeta(...)$), which firms rationally forecast in t-1, and random noise (i.e., u_{ft} and ξ_{ft}), which they do not observe in t-1. These random terms are by construction mean independent from $\beta(...)$ and $\zeta(...)$ and capture uncertainty in productivity growth.

Empirical Strategy. Taking the logarithm of the ratio between the first-order conditions with respect to M_{ft} and L_{ft} in equation (5) delivers:

$$\log\left(\frac{M_{ft}}{L_{ft}}\right) = \sigma \log\left(\frac{w_{ft}}{r_{ft}}\right) - \log\left(B_{ft}\right).$$
(8)

⁴⁶This assumption parallels the standard learning-by-doing idea in the production estimation literature (De Loecker, 2013; Doraszelski and Jaumandreu, 2013, 2018).

⁴⁷I treat physical inputs and labor as the only static inputs for simplicity. Doraszelski and Jaumandreu (2018) show that the model can be extended, at computational costs, to feature, e.g., dynamic inputs or substitutability between high-skilled and low-skilled labor.

⁴⁸Humlum (2021) shows that equation (5) can be micro-funded using a task-based approach of firm production, which is the standard way to model robot adoption in the literature (Acemoglu and Restrepo, 2018a,b; Graetz and Michaels, 2018; Acemoglu and Restrepo, 2020).

Equation (8) states that the firm-level physical inputs-to-labor ratio depends upon (i) relative factor prices, (ii) elasticity of substitution, and (iii) labor-augmenting productivity. An increase of w_{ft}/r_{ft} raises the physical inputs-to-labor employment ratio only if $\sigma > 1$ (gross substitutes), and it decreases it if $\sigma \in (0, 1)$ (gross complements). A decrease of B_{ft} always shifts the physical inputs-to-labor upwards. Substituting equation (6) inside equation (8) delivers:

$$\log\left(\frac{M_{ft}}{L_{ft}}\right) = \sigma \log\left(\frac{w_{ft}}{r_{ft}}\right) - \beta \left(\log\left(B_{ft-1}\right), R_{ft-1}, O_{ft-1}\right) - u_{ft}.$$
(9)

Estimating factor-biased productivity in equation (9) requires addressing two challenges. First, r_{ft} and B_{ft-1} are unobserved. Second, although $\beta(...) \perp u_{ft}$ by definition, relative input prices may still be correlated with u_{ft} . I address the first issue by assuming that physical inputs is perfectly mobile across firms within an industry, i.e., $r_{ft} = r_{jt}$.⁴⁹ I obtain an expression for $\log(B_{ft-1})$ by recursively substituting equation (8) inside equation (9). Then, I approximate the unknown function $\beta(...)$ with the following polynomial:⁵⁰

$$\beta(\dots) \approx \underbrace{\sum_{s=0}^{4} \left(\log\left(\frac{M_{ft-1}}{L_{ft-1}}\right) - \sigma \log\left(\frac{w_{ft-1}}{r_{jt-1}}\right) \right)^s}_{= \log(B_{ft-1})} + \beta_1 R_{ft-1} + \beta_2 O_{ft-1}.$$
(10)

Equation (10) clarifies that σ enters non-linearly equation (9) via $\log(B_{ft-1})$. β_1 and β_2 measure the semi-elasticity of factor-specific productivity to past robot choices and ownership structure.

I address the second issue by instrumenting $\log (w_{ft}/r_{jt})$ with $\log(L_{ft-1})$, i.e., past employment choices. Doraszelski and Jaumandreu (2018) show that this is a valid (excluded) instrument under the assumption that factor-biased productivity follows a first-order Markov process. I jointly estimate σ , β_1 , and β_2 using a just-identified GMM estimator (Hansen, 1982). The three moment conditions I use are $\mathbb{E}[\log(L_{ft-1})u_{ft}] = 0$, $\mathbb{E}[R_{ft-1}u_{ft}] = 0$, and $\mathbb{E}[O_{ft-1}u_{ft}] = 0$. With a consistent estimate of σ and B_{ft} , I recover TFP by inverting equation (5):

$$\log(\widehat{z}_{it}) = \log(y_{ft}) - \frac{\widehat{\sigma}}{\widehat{\sigma} - 1} \log\left(M_{ft}^{\frac{\widehat{\sigma} - 1}{\widehat{\sigma}}} + \widehat{B}_{ft}^{\frac{1}{\widehat{\sigma}}} L_{ft}^{\frac{\widehat{\sigma} - 1}{\widehat{\sigma}}}\right).$$
(11)

⁴⁹The assumption that physical inputs are perfectly mobile across firms is standard when estimating factor-biased productivity (Raval, 2019; Oberfield and Raval, 2021).

⁵⁰Polynomial approximations of this type are standard in the production function estimation literature (Olley and Pakes, 1996; Levinsohn and Petrin, 2003; Doraszelski and Jaumandreu, 2013, 2018).

Since I deflate firms' sales with a Paasche-type output price change index, it measures quantity-based TFP. Finally, to identify the contribution of foreign-ownership and robot adoption on firm-level Hicks-neutral productivity, I parametrize $\log(z_{it})$ as follows:

$$\log(\hat{z}_{ft}) = \zeta_0 + \zeta_1 \log(\hat{z}_{ft-1}) + \zeta_2 R_{ft-1} + \zeta_3 O_{ft-1} + \xi_{ft}.$$
 (12)

Sample Selection. Bringing the model to the data requires addressing one last issue. Namely, dealing with the remaining missing values in the cleaned sample I create in Section 3.2. I address this issue as follows. I bin the data into seven mutually exclusive time intervals [1993-1995], [1996-1998], [1999-2001], [2002-2004], [2005-2007], [2008-2010], and [2011-2014]. If a firm is foreign-owned at least once in a given time interval, I set $O_{ft} = 1$. If not, I set $O_{ft} = 0$. Similarly, If a firm employs a robot at least once in a given time interval, I set $R_{ft} = 1$. If not, I set $R_{ft} = 1$. If not, I set $R_{ft} = 1$. If not, I set $R_{ft} = 0$. I take the time average within each interval for the continuous variables (physical inputs, labor, sales, and wages) disregarding missing values.

6.2 Estimation Results

Table 5 shows the estimates I obtain from equation (9). For benchmarking purposes, Column (1) and (2) report the OLS, respectively GMM, estimates when setting $\beta_1 = \beta_2 = 0$. The OLS estimator delivers an elasticity of substitution equal to 1.30, which leads to the conclusion that physical inputs and labor are gross substitutes. By contrast, the GMM estimate equals 0.75, implying that they are gross complements. These estimates are in line with existing literature (Doraszelski and Jaumandreu, 2018; Raval, 2019; Humlum, 2021; Oberfield and Raval, 2021). Even if estimating the elasticity of substitution between physical inputs and labor is not the primary goal of my analysis, replicating the findings of previous studies increases the estimator credibility.

In Column (3), I estimate equation (9) allowing only past productivity and robot adoption choices to impact current labor-augmenting productivity (i.e., setting $\beta_2 = 0$). Column (3) delivers two main insights. First, it confirms that physical inputs and labor are gross complements at the firm-level. Second, it shows that robot adopters have lower labor-augmenting productivity and a 40% higher physical inputs-to-labor ratio than non-adopters, on average.⁵¹ Finally, Column (4) reports the estimates from the full model. Conditional on input prices and robot adoption, the estimates reveal that past ownership structure does not produce a direct significant effect on current labor-

⁵¹The estimated coefficient of β_1 is -0.33, i.e., robot adoption reduces labor-augmenting productivity. However, because B_{ft} enters with negative sign in equation (8), the coefficient in Table 5 is positive. Finally, since the left-hand side of equation (8) is in logarithm, the average effect is $(\exp(0.33) - 1) = 40\%$. The same argument applies to the estimated coefficient of β_2 .

augmenting productivity.⁵² Overall, the estimates suggest that it is robot adoption, rather than foreign ownership per se, to tilt production towards physical inputs.

Dependent Variable:	$Log(Physical Inputs/Employees)_{ft}$			
	(1)	(2)	(3)	(4)
$Log(Wage/Int. Rate)_{ft}$	1.3^{***} (0.04)	0.75^{***} (0.05)	0.65^{***} (0.05)	0.67^{***} (0.06)
$\operatorname{Robot}_{ft-1}$		~ /	0.33^{***} (0.05)	0.34^{***} (0.05)
$\operatorname{Foreign-Owned}_{ft-1}$				-0.13 (0.13)
Observations Estimator	5,697 OLS	5,697 GMM	5,697 GMM	5,697 GMM

Table 5: LABOR-AUGMENTING PRODUCTIVITY

Note: The unit of observation is a firm-time interval pair. A time interval is a three-year time window. $Log(Physical Inputs/Employees)_{ft}$ is the log of the ratio of the physical input expenditure and number of employees. $Log(Wage/Int. Rate)_{ft}$ is the log of the ratio of the gross workers' wages and capital interest rate. $Robot_{ft}$ is a binary variable equal to 1 if firm f employs a robot in the current time interval and 0 otherwise. Foreign-Owned_{ft} is a binary variable equal to 1 if firm f is foreign-owned in the current time interval and 0 otherwise. Standard errors in parenthesis. Significance levels: *** 0.01, ** 0.05, * 0.1.

Because the model is just-identified, I cannot test if the exclusion restrictions jointly hold. However, following the method proposed by Andrews, Gentzkow and Shapiro (2017), Figure D.1 shows that the estimates in Column (4) are insensitive to local deviations in the moment condition involving the excluded instrument (i.e., $\mathbb{E} \left[\log(L_{ft-1})u_{ft} \right] =$ 0). This fact increases the confidence that past employment level are a valid instrument for current relative factor prices and that the parameters of interest are identified.

Next, I estimate TFP from quation (12). Table 6 shows the estimates I obtain by adding one regressor at a time.⁵³ Column (1) shows that TFP is strongly persistent over time, with the coefficient associated with the auto-regressive term being equal to 0.97. Column (2) shows that, conditional on past productivity, robot adopters are about 32% more productive than non-adopting firms, on average. Finally, Column (3) shows that, conditional on past productivity and robot adoption, foreign-owned firms are

⁵²However, because it spurs robot adoption (see Table 2), foreign ownership does produce an indirect effect on within-firm input factors' reallocation.

⁵³In each column, I also include the current value of the log of labor-augmenting productivity to purge the indirect effect that robot adoption and foreign ownership produce on TFP through it. The unconditional correlation between Log TFP and Log labor-augmenting productivity is about 0.95. In an unreported specification, I also compute bootstrap standard errors to account for the fact that TFP is estimated. The significance levels in Table 6 are unaltered.

about 43% more productive than domestic ones.⁵⁴ Hence, unlike for labor-augmenting productivity, both robot adoption and foreign ownership contribute to TFP growth.

Dependent Variable:		$Log(TFP)_{ft}$	
	(1)	(2)	(3)
$Log(TFP)_{ft-1}$	0.97^{***}	0.96***	0.96***
	(0.002)	(0.002)	(0.002)
$\operatorname{Robot}_{ft-1}$		0.32^{***}	0.29^{***}
		(0.05)	(0.05)
Foreign-Owned $_{ft-1}$			0.43^{***}
			(0.15)
Observations	3,442	3,442	3,442
Estimator	OLS	OLS	OLS

Table 6: TOTAL FACTOR PRODUCTIVITY

Note: The unit of observation is a firm-time interval pair. A time interval is a three-year time window. $\text{Log}(\text{TFP})_{ft}$ is the log of total factor productivity. Robot_{ft} is a binary variable equal to 1 if firm f employs a robot in the current time interval and 0 otherwise. Foreign-Owned_{ft} is a binary variable equal to 1 if firm f is foreign-owned in the current time interval and 0 otherwise. Standard errors in parenthesis. Significance levels: *** 0.01, ** 0.05, * 0.1.

6.3 Counterfactuals

I use equations (10) and (12) to infer the relative importance of foreign ownership and robot adoption to explain changes in industry-level TFP and the labor share in the Spanish manufacturing industry over the sample period. Simulating these equations forward while shutting down the contribution of robot adoption or foreign ownership on productivity allows me to reconstruct counterfactual firm-level productivity paths, which I then aggregate up at the industry level.

I consider two different counterfactual scenarios. In the first one, I shut down the effect of foreign ownership on productivity (i.e., I set $\zeta_3 = 0$ in equation (12)). To account for the effect that foreign ownership exerts on the probability of robot adoption, I also discount ζ_2 in equation (12) by 10%, which is the foreign ownership robot adoption premium I estimate in Table 2. In the second one, I shut down both the effect of foreign ownership and robot adoption (i.e., I set $\beta_1 = 0$ in equation (10) and $\zeta_2 = \zeta_3 = 0$ in equation (12)). Since it is insignificant in Table 5, I set $\beta_2 = 0$ in all scenarios.

⁵⁴This result is consistent with literature showing that foreign-owned firms are better managed than domestic ones (Bloom et al., 2012).

In each scenario, I compute counterfactual outcomes in two steps. First, starting from their observed productivity levels in the first sample period, I simulate counterfactual firm-level TFP and labor share paths by rolling forward equations (10) and (12).⁵⁵ Second, I aggregate the results at the industry level using the observed firms' employment shares each year, which I keep constant across scenarios. To avoid dealing with sample attrition due to entry and exit, I focus on the sub-sample of active firms between 1993 and 2014.⁵⁶ I provide further implementation details in Appendix D.2.

Figure 2 shows the results. For benchmark purposes, the first scenario (Baseline) shows the observed industry-level TFP and labor share trajectories. The results indicate that TFP grew by about 0.60% and labor share decreased by about 18.07% over the sample period. These figures are qualitatively similar to Doraszelski and Jaumandreu (2018). The second scenario (No MNEs) shows the counterfactual outcomes that I obtain when shutting down the effect of foreign ownership. Absent MNEs, the total TFP change would have been equal to -0.25%, i.e., 0.85 percentage points lower than the baseline. By contrast, the labor share would have decreased by 16.60%, i.e., 1.47 percentage points less than under the baseline scenario.

The third scenario (Neither MNEs nor Robots) shows the counterfactual outcomes that I obtain when shutting down the foreign ownership and robot adoption effects on productivity simultaneously. In the absence of MNEs and robots, the total TFP change would have been -0.50%, i.e., about 1.10 percentage point lower than the baseline. By contrast, the labor share would have increased by 1.20%, i.e., about 19.27 percentage points more than the baseline. Hence, multinational activity explain ($0.85/1.10 \approx$) 77% of the total TFP drop between the baseline and a scenario without MNEs and robots. By contrast, it explains about ($1.47/19.27 \approx$) 7% of the total labor share increase across counterfactual scenarios.

Overall, the counterfactual exercise delivers two insights. First, multinational activity produces substantial TFP gains but puts downward pressure on the labor share by boosting robot adoption. However, the first effect is one order of magnitude larger than the second one.⁵⁷ Although the model is silent about total welfare, the results provide new evidence about the efficiency versus equity trade-off that policymakers face when

⁵⁵Firm-level TFP comes from equation (11). The labor (cost) share is $w_{ft}L_{ft}/(w_{ft}L_{ft}+r_{jt}M_{ft})$.

⁵⁶There are 161 firms in this sub-sample. 5 become foreign-owned, whereas 70 employ a robot at least once. By contrast, there are 3611 firms in the baseline sample. 123 become foreign-owned, whereas 874 employ a robot at least once. While the share of foreign-owned firms is about 3% in both samples, the share of robot adopters in the sub-sample (about 43%) is about twice as large than in the baseline sample (about 24%). Therefore, the results I obtain when shutting down the robot adoption productivity effect should be understood as an upper bound of the effect in the full sample.

⁵⁷This number mirrors the fact that foreign ownership produces both a direct and an indirect effect (via robot adoption) on TFP (Table 6) but only an indirect one on labor-augmenting productivity (Tables 2 and 5).



Figure 2: Counterfactual Scenarios

Note: The Figure shows the changes of industry-level TFP and labor share over the sample period under three different scenarios. The first scenario reports total growth rates under the baseline scenario. In this scenario, the total TFP growth rate is 0.60% and the total labor share growth rate is -18.07%. The second scenario is a counterfactual without MNEs. In this case, I also discount the robot adoption productivity effect of foreign-owned firms by 10%, which is the foreign ownership robot adoption premium I estimate in Table 2. In this scenario, the total TFP growth rate is -0.25% and the total labor share growth rate is -16.60%. Finally, the third scenario is a counterfactual without MNEs and robots. In this scenario, the total TFP growth rate is -0.50% and the total labor share growth rate is 1.20%. See Appendix D.2 for the implementation details.

attracting foreign investment.

Second, the results offer a new perspective on the observed decline in the manufacturing labor share in many advanced economies. In a recent survey of the literature, Grossman and Oberfield (forthcoming) mention globalization and automation as two leading explanations for this phenomenon. Figure 2 reinforces and extends their argument. Rather than being separate causes, the joint interaction of MNEs' activities (globalization) and robot adoption (technological change) contributes the observed decline. Notwithstanding, robots are disproportionally more important than multinationals in driving this process.

7 Conclusion

In this paper, I show that multinational enterprises spur the adoption of industrial robots. This result is new in the literature and has relevant policy implications because robots may increase productivity but also replace workers in routine tasks.

I exploit new cross-country industry-level data and detailed firm-level data for Spanish manufacturing. I use the first data source to document a positive and robust correlation between multinational production and robot adoption. I use the second data source to study if firms switching from domestic to foreign ownership become more likely to employ robots. Combining a difference-in-differences approach with a propensity score reweighing estimator, I find that acquired firms are about 10% more likely to employ robots. In terms of mechanism, I provide evidence that foreign parents grant increased market access to their affiliates. In turn, the possibility of expanding into global markets creates incentives to scale-up production, and robot adoption is one way to accomplish this result.

Whereas switching to foreign ownership generates significant gains for their affiliates (reflected, e.g., in an increase in sales and employment), I also find evidence that the labor share falls within acquired firms, indicating that switching to foreign ownership shifts income away from labor. These insights inform a dynamic partial equilibrium firm investment model, which I use to further investigate the productivity and distributional effects of foreign ownership and robot adoption.

The model estimates deliver two main insights. First, robot adoption and foreign ownership increase firm-level total factor productivity. Second, it is robot adoption, rather being foreign-owned per se, to reallocate income away from labor. A counterfactual economy without robots and multinationals would have lower productivity but a higher labor share. Overall, these results provide new evidence on the efficiency versus equity trade-off that policymakers face when attracting foreign investment.
References

- Acemoglu, Daron and Pascual Restrepo, "Modeling Automation," AEA Papers and Proceedings, 2018.
- **and** _ , "The Race between Man and Machine: Implications of Technology for Growth, Factor Shares, and Employment," *American Economic Review*, 2018.
- _ and _ , "Robots And Jobs: Evidence From US Labor Markets," Journal of Political Economy, 2020.
- _, Claire Lelarge, and Pascual Restrepo, "Competing with Robots: Firm-Level Evidence from France," AEA Papers and Proceedings, 2020.
- Aghion, Philippe, Céline Antonin, Simon Bunel, and Xavier Jaravel, "What Are the Labor and Product Market Effects of Automation? New Evidence from France," 2020.
- Aitken, Brian J. and Ann E. Harrison, "Do Domestic Firms Benefit from Direct Foreign Investment? Evidence from Venezuela," *American Economic Review*, 1999.
- Alfaro-Ureña, Alonso, Isabela Manelici, and Jose P. Vasquez, "The Effects of Multinationals on Workers: Evidence from Costa Rican Microdata," *Manuscript*, 2021.
- _ , _ , and _ , "The Effects of Joining Multinational Supply Chains: New Evidence from Firm-to-Firm Linkages," *The Quarterly Journal of Economics*, forthcoming.
- Andrews, Isaiah, Matthew Gentzkow, and Jesse M. Shapiro, "Measuring the Sensitivity of Parameter Estimates to Estimation Moments," *The Quarterly Journal* of Economics, 2017.
- Anelli, Massimo, Italo Colantone, and Piero Stanig, "We Were the Robots: Automation and Voting Behavior in Western Europe," BAFFI CAREFIN Centre Research Paper, 2019.
- Antras, Pol and Elhanan Helpman, "Global Sourcing," Journal of Political Economy, 2004.
- Artuc, Erhan, Bastos Paulo, and Bob Rijkers, "Robots, Tasks and Trade," World Bank Working Paper, 2018.

- Autor, David, David Dorn, Lawrence F. Katz, Christina Patterson, and John Van Reenen, "The Fall of the Labor Share and the Rise of Superstar Firms," *The Quarterly Journal of Economics*, 2020.
- Baldwin, Richard, "The Globotics Upheaval: Globalization, Robotics, and the Future of Work," Oxford University Press, 2019.
- Bilir, L. Kamran and Eduardo Morales, "Innovation in the Global Firm," *Journal* of Political Economy, 2020.
- **Bircan, Cagatay**, "Ownership Structure and Productivity of Multinationals," *Journal* of International Economics, 2019.
- Blalock, Garrick and Paul J. Gertler, "Welfare Gains from Foreign Direct Investment through Technology Transfer to Local Suppliers," *Journal of International Economics*, 2008.
- Bloom, Nicholas, Christos Genakos, Raffaella Sadun, and John Van Reenen, "Management Practices across Firms and Countries," *Academy of Management Perspectives*, 2012.
- Blundell, Richard and Monica Costa Dias, "Alternative Approaches to Evaluation in Empirical Microeconomics," *Journal of Human Resources*, 2009.
- Borusyak, Kirill, Xavier Jaravel, and Jann Spiess, "Revisiting Event Study Designs: Robust and Efficient Estimation," *arXiv preprint arXiv:2108.12419*, 2021.
- Branstetter, Lee G., Raymond Fisman, and C. Fritz Foley, "Do Stronger Intellectual Property Rights Increase International Technology Transfer? Empirical Evidence from US Firm-Level Panel Data," *The Quarterly Journal of Economics*, 2006.
- Brown, James R., Gustav Martinsson, and Bruce C. Petersen, "Do Financing Constraints Matter for R&D?," *European Economic Review*, 2012.
- Brynjolfsson, Erik and Andrew McAfee, The Second Machine Age: Work, Progress, and Prosperity in a Time of Brilliant Technologies, New York: W. W. Norton, 2014.
- **Bustos, Paula**, "Trade Liberalization, Exports, and Technology Upgrading: Evidence on the Impact of MERCOSUR on Argentinian Firms," *American Economic Review*, 2011.

- Callaway, Brantly and Pedro H.C. Sant'Anna, "Difference-in-Differences with Multiple Time Periods," *Journal of Econometrics*, 2021.
- Chaisemartin, Clément De and Xavier d'Haultfoeuille, "Two-Way Fixed Effects Estimators with Heterogeneous Treatment Effects," American Economic Review, 2020.
- DiNardo, John, Nicole Fortin, and Thomas Lemieux, "Labor Market Institutions and the Distribution of Wages, 1973-1992: A Semiparametric Approach," *Econometrica*, 1996.
- **Doraszelski, Ulrich and Jordi Jaumandreu**, "R&D and Productivity: Estimating Endogenous Productivity," *Review of Economic Studies*, 2013.
- and _ , "Measuring the Bias of Technological Change," Journal of Political Economy, 2018.
- Feenstra, Robert C. and Gordon H. Hanson, "Foreign Direct Investment and Relative Wages: Evidence from Mexico's Maquiladoras," *Journal of International Economics*, 1997.
- Freeman, Richard B., "Who Owns the Robots Rules the World," *IZA World of Labor*, 2015.
- Goodman-Bacon, Andrew, "Difference-in-Differences with Variation in Treatment Timing," Journal of Econometrics, 2021.
- Graetz, Georg and Guy Michaels, "Robots at Work," *Review of Economics and Statistics*, 2018.
- Grossman, Gene M. and Ezra Oberfield, "The Elusive Explanation for the Declining Labor Share," Annual Review of Economics, forthcoming.
- Guadalupe, Maria, Olga Kuzmina, and Catherine Thomas, "Innovation and Foreign Partnership," *American Economic Review*, 2012.
- Hansen, Lars Peter, "Large Sample Properties of Generalized Method of Moments Estimators," *Econometrica*, 1982.
- Harrison, Ann, Globalization and Poverty, University of Chicago Press, 2007.
- and Andrés Rodríguez-Clare, "Trade, Foreign Investment, and Industrial Policy for Developing Countries," *Handbook of Development Economics*, 2010.

- Harrison, Ann E. and Margaret S. McMillan, "Does Direct Foreign Investment Affect Domestic Credit Constraints?," *Journal of International Economics*, 2003.
- Hémous, David and Morten Olsen, "The Rise of the Machines: Automation, Horizontal Innovation, and Income Inequality," *American Economic Journal: Macroeconomics*, 2022.
- Humlum, Anders, "Robot Adoption and Labor Market Dynamics," *Working Paper*, 2021.
- International Federation of Robotics, "World Robotics Industrial Robots," Technical Report 2019.
- **International Monetary Fund**, "Fifth Edition of the International Monetary Fund's Balance of Payments Manual," Technical Report 2007.
- Jardet, Caroline, Cristina Jude, and Menzie D. Chinn, "Foreign Direct Investment under Uncertainty: Evidence from a Large Panel of Countries," NBER Working Paper, 2022.
- Javorcik, Smarzynska Beata, "Does Foreign Direct Investment Increase The Productivity of Domestic Firms? In Search of Spillovers Through Backward Linkages," *American Economic Review*, 2004.
- Keller, Wolfgang, "Knowledge Spillovers, Trade, and Foreign Direct Investment," Technical Report, National Bureau of Economic Research 2021.
- and Stephen Ross Yeaple, "The Gravity of Knowledge," American Economic Review, 2013.
- Koch, Michael and Marcel Smolka, "Foreign Ownership and Skill-Biased Technological Change," *Journal of International Economics*, 2019.
- _, Ilya Manuylov, and Marcel Smolka, "Robots and Firms," The Economic Journal, 2021.
- Korinek, Anton and Joseph E. Stiglitz, "Artificial Intelligence, Globalization, and Strategies for Economic Development," *NBER Working Paper*, 2021.
- Levinsohn, James and Amil Petrin, "Estimating Production Functions using Inputs to Control for Unobservables," *Review of Economic Studies*, 2003.

- Lileeva, Alla and Daniel Trefler, "Improved Access to Foreign Markets Raises Plant-Level Productivity... for Some Plants," *The Quarterly Journal of Economics*, 2010.
- Loecker, Jan De, "Detecting Learning by Exporting," American Economic Journal: Microeconomics, 2013.
- _, Jan Eeckhout, and Gabriel Unger, "The Rise of Market Power and the Macroeconomic Implications," The Quarterly Journal of Economics, 2020.
- Manova, Kalina, Shang-Jin Wei, and Zhiwei Zhang, "Firm Exports and Multinational Activity under Credit Constraints," *Review of Economics and Statistics*, 2015.
- Méndez-Chacón, Esteban and Diana Van Patten, "Multinationals, Monopsony and Local Eevelopment: Evidence from the United Fruit Company," *Manuscript*, 2021.
- **Oberfield, Ezra and Devesh Raval**, "Micro Data and Macro Technology," *Econometrica*, 2021.
- Olley, Steven and Ariel Pakes, "The Dynamics of Poductivity in the Telecommunications Equipment Industry," *Econometrica*, 1996.
- Raval, Devesh, "The Micro Elasticity of Substitution and Non-Neutral Technology," The RAND Journal of Economics, 2019.
- Setzler, Bradley and Felix Tintelnot, "The Effects of Foreign Multinationals on Workers and Firms in the United States," *The Quarterly Journal of Economics*, 2021.
- Stapleton, Katherine and Michael Webb, "Automation, Trade and Multinational Activity: Micro Evidence from Spain," SSRN Working Paper, 2020.
- Sun, Chang, "Multinational Production With Non-Neutral Technologies," Journal of International Economics, 2020.
- Sun, Liyang and Sarah Abraham, "Estimating Dynamic Treatment Effects in Event Studies with Heterogeneous Treatment Effects," *Journal of Econometrics*, 2021.
- UNCTAD, "World Investment Report," Technical Report 2013.
- _, "World Investment Report," Technical Report 2018.

Appendices

Appendix A Firm-Level Data Description

A.1 Tables

	Don	Domestic		eign
	Mean	SD	Mean	SD
Panel A: Automation Technology				
Robot	0.17	0.37	0.36	0.48
Numerically Controlled Machines	0.39	0.49	0.55	0.50
CAD Manufacturing	0.28	0.45	0.44	0.50
Flexible Systems	0.19	0.40	0.39	0.49
Panel B: Type of Manufacturing				
Batch Manufacturing	0.55	0.50	0.24	0.43
Mass Manufacturing	0.35	0.48	0.61	0.49
Continuous Manufacturing	0.08	0.26	0.14	0.34
Mixed Manufacturing	0.02	0.14	0.02	0.13
Panel C: Technological Effort				
Effort to Assimilate Foreign Technology	0.06	0.24	0.24	0.43
Import of Foreign Technology (Hundred Euros)	0.05	1.17	2.41	14.45
Panel D: Innovation and Research and Development				
Innovation Plan	0.09	0.29	0.39	0.49
Total RD Expenses (Million Euros)	0.08	1.43	1.41	3.40
Internal RD (Million Euros)	0.05	0.91	1.02	2.55
Panel E: Other Characteristics				
Exporter	0.47	0.50	0.86	0.35
Export Via Parental Network	0.00	0.00	0.29	0.45
Exports (Million Euros)	2.50	17.02	41.32	122.29
No. of Export Markets	0.46	0.84	1.20	1.09
Gross Wage per Employee (Thousand Euros)	24.78	9.30	35.95	12.91
Employees	64.86	202.19	594.33	1283.18
Share of Graduates	0.05	0.09	0.08	0.08
Investment (Million Euros)	0.31	1.44	4.80	23.59
Physical Inputs (Million Euros)	5.12	20.86	115.64	428.56
Sales (Million Euros)	10.06	46.91	141.42	421.64
Labor Productivity (Million Euros)	0.12	0.17	0.18	0.15

Table A.1: Summary Statistics by Ownership Type

Note: The Table shows the average (Mean) and standard deviation (SD) of the ESEE variables in my sample. The sub-sample of domestic firms includes those that stay under domestic ownership during their life span. The sub-sample of foreign-owned firms includes those that switch from domestic to foreign ownership at most once during their life span. Overall, foreign-owned firms outperform domestic-owned ones in all dimension. See Table A.2 for the variables' description.

Variable	Range/Unit	Frequency	Description
Robot Adoption	[0, 1]	Q	= 1 if firm employs robot
Numerically Controlled Machines	[0, 1]	Q	= 1 if firm employs numerically controlled machines
CAD Manufacturing	[0, 1]	Q	= 1 if firm employs CAD manufacturing
Flexible Systems	[0, 1]	Q	= 1 if firm employs flex. systems
Batch Manufacturing	[0, 1]	Q	= 1 if firm performs batch manuf.
Mass Manufacturing	[0, 1]	Q	= 1 if firm performs mass manuf.
Continuous Manufacturing	[0, 1]	Q	= 1 if firm performs continuous manuf.
Mixed Manufacturing	[0, 1]	Q	= 1 if firm performs mixed manuf.
Import of Foreign Technology	Euros	Q	Payments for licenses and technical aid from abroad
Total RD Expenses	Euros	А	Total research and development expenses
Internal RD	Euros	А	Internal research and development expenses
Exporter	[0, 1]	А	= 1 If firm exports abroad
Export Via Parental Network	[0, 1]	Q	= 1 If firm exports via its foreign parent
Exports	Euros	А	Value of exports
No. of Export Markets	$[0,\infty)$	А	Number of foreign markets served
Gross Wage per Employee	Euros	А	Labor costs divided by number of employees
Employees	$[0,\infty)$	А	Total number of employees
Investment	Euros	А	Value of investment in tangible assets
Physical Inputs	Euros	А	Value ot tangible fixed assets (no buildings and land)
Sales	Euros	А	Value of sales
Labor Productivity	Euros	А	Sales per employee
Prince Index	[-100, 100]	А	Change of the sale prices between two years
Foreign-Owned	Euros	А	= 1 If firm is foreign-owned
Owns Equity Shares Abroad	Euros	A (post 2000)	= 1 If firms controls foreign companies
Activity	Number	А	2-digit manufacturing industry
Change Nature	Factor	А	Reports if firm changed its nature

Table A.2: VARIABLES' DESCRIPTION

Note: The Table shows name, range or unit, frequency, and description of the ESEE variables I use in my analysis. A stands for "annual" and Q for "quadrennial".

A.2 Figures



Figure A.1: COVARIATES' DENSITY PLOTS BY OWNERSHIP

Note: The Figure shows the empirical probability density function (pdf) of the log of employees, sales, physical inputs, R&D expenses, wages, and investment by ownership type. I estimate the empirical pdf functions for domestic-owned firms based on their lifetime characteristics. I estimate it only for the years before the acquisition date for foreign-owned ones. Overall, foreign-owned firms outperform domestic-owned ones along all dimensions already before the acquisition.



Figure A.2: Share of Robot Adopters by Ownership and Year

Note: The Figure shows the share of robot adopters among foreign-owned and domestic-owned firms between 1994 and 2014. Although the share of robot adopters has steadily grown in both sub-samples, the proportion of robot adopting firms among foreign-owned firms is about twice as large than among domestic-owned ones in any given year.



Figure A.3: Share of Robot Adopters by Ownership and Industry

Domestic Foreign

Note: The Figure shows the share of robot adopters among foreign-owned and domestic-owned firms across industries. The shares are averages between 1994 and 2014. Overall, the share of robot adopting firms among foreign-owned firms is systematically larger than among domestic-owned ones across industries. The automotive industry (C29) witnesses the highest adoption rates.

Appendix B Industry-Level Data Description

B.1 Sample Construction and Data Cleaning

Using the IFR data requires addressing two issues. First, when constructing the stock of robots, the IFR assumes a depreciation rate of zero for the first twelve years of service. After that, they assume full depreciation. This data feature contrasts with standard notions of physical capital accumulation. For this reason, I follow Graetz and Michaels (2018) and employ a permanent inventory method to compute the stock of robots in each country-industry-year cell. The procedure consists of two steps. In the first step, I take as initial value the first available data about the number of deployed robots at the country-industry level. In the second one, I calculate subsequent values using the information about new installations and assuming a yearly depreciation rate of 10%.

Second, about 20% of the stock cannot be allocated to any ISIC review 4 industry in some countries. I follow Graetz and Michaels (2018) and allocate these robots proportionally to each sector based on their share of deployed robots across all sample years. I present my baseline results using the adjusted data. However, the my findings in Section 4 hold regardless of this choice.

Besides, merging data from AMNE, IFR, and WIOD SEA also requires tackling two challenges. First, one has to homogenize industry definitions. AMNE and WIOD follow indeed the ISIC review 4 classification, whereas the IFR has its own system. However, since the IFR closely follows the ISIC review 4, it is feasible to match industries without ambiguity based on the industry description.

Second, the three datasets use a different industry aggregation level. Because the AMNE data have the most aggregate industry classification, I group industries in the IFR and WIOD SEA to match the AMNE classification. The final dataset contains the following industries: "A" (Agriculture, forestry and fishing), "B" (Mining and quarrying), "C1012" (Manufacture of food products, beverages and tobacco products), "C1315" (Manufacture of textiles, wearing apparel, leather and related products). "C16" (Manufacture of wood and of products of wood and cork, except furniture; manufacture of articles of straw and plaiting materials), "C1718" (Manufacture of paper and paper products, printing and reproduction of recorded media), "C19" (Manufacture of coke and refined petroleum products), "C2021" (Manufacture of chemicals chemical products, pharmaceuticals, medicinal chemical and botanical products), "C22" (Manufacture of rubber and plastics products), "C23" (Manufacture of other non-metallic mineral products), "C24" (Manufacture of basic metals), "C25" (Manufacture of fabricated metal products, except machinery and equipment), "C26" (Manufacture of computer, electronic and optical products), "C27" (Manufacture of electrical equipment), "C28" (Manufacture of machinery and equipment n.e.c.), "C29" (Manufacture of motor

vehicles, trailers and semi-trailers), "C30" (Manufacture of other transport equipment), "DE" (Electricity, gas, steam and air conditioning supply), "F" (Construction), "P" (Education and R&D).

I deflate nominal variables using the one-digit domestic output deflator provided by WIOD SEA. I express them in previous-year prices using the following formula:

$$X_{pyp}(t) = X(t) \times \frac{D(t)}{D(t-1)}$$

 $X_{pyp}(t)$ is the variable of interest in year t in previous-year prices, X(t) is the nominal level of the variable of interest and D(t)/D(t-1) is the ratio of the output deflator between two consecutive periods. Finally, I convert them into US dollars using the nominal exchange rate provided by WIOD SEA.

The final dataset includes the following countries: Australia, Austria, Belgium, Bulgaria, Brazil, Switzerland, China, Czech Republic, Germany, Denmark, Spain, Estonia, Finland, France, Great Britain, Greece, Croatia, Hungary, Indonesia, India, Ireland, Italy, Japan, South Korea, Lithuania, Latvia, The Netherlands, Norway, Poland, Portugal, Romania, Russia, Slovakia, Sweden, Slovenia, Tukey and the USA.

B.2 Additional Results

Variable	Ν	Mean	St. Dev.	Min	Q25	Median	Q75	Max
Log Robot Stock	6563	2.95	3.49	-8.21	0.95	3.12	5.29	12.75
Log MNEs' Production	6563	7.64	2.02	-0.05	6.39	7.80	9.02	12.78
Log Employees	6563	4.61	2.14	-3.22	3.18	4.47	5.91	12.60
Log Capital Stock	6563	9.27	2.07	1.11	8.05	9.28	10.74	14.63
Log Wages	6563	7.94	1.93	0.85	6.76	7.89	9.27	13.73
Log Interest Rate	6563	7.58	2.03	-1.46	6.39	7.62	8.87	12.79

Table B.1: SUMMARY STATISTICS

Note: The Table shows summary statistics for the sample I construct in section 3.1.

B.3 Tables

Dependent Variable:	Log(Robots (1)	/Thousand Employees) _{cit} (2)
Log(Multinational Production) _{cit}	0.22***	0.26***
	(0.03)	(0.03)
$Log(Capital Stock/Thousand Employees)_{cit-1}$	0.15^{***}	0.13^{***}
	(0.04)	(0.04)
$Log(Interest Rate/Wages)_{cit-1}$	-0.09*	-0.08*
	(0.04)	(0.04)
Industry-Year FE	Yes	Yes
Country-Year FE	Yes	Yes
Sample	No $C29$	No Top 5
Observations	$5,\!429$	4,886
Estimator	OLS	OLS

Table B.2: INDUSTRY-LEVEL CORRELATIONS (ROBUSTNESS)

Note: The unit of observation is a country-industry-year tuple. $\text{Log}(\text{Robots}/\text{Thousand Employees})_{cit}$ is the log of the number of robots per thousand employees in country c's industry i in year t. $\text{Log}(\text{Multinational Production})_{cit}$ is the log of gross output produced by foreign-owned firms in country c's industry i in year t. $\text{Log}(\text{Capital Stock}/\text{Thousand Employees})_{cit}$ is the capital stock per thousand employees in country c's industry i in year t. $\text{Log}(\text{Interest Rate}/\text{Wages})_{cit}$ is the ratio between the capital interest rate and gross wages in country c's industry i in year t. Heteroscedasticity-robust standard errors in parenthesis. Significance levels: *** 0.01, ** 0.05, * 0.1.

B.3.1 Specification Robustness

To ensure that the baseline results do not merely pick-up pre-sample trends, I estimate the following long-difference equation:

$$\Delta R_{ci} = \beta \Delta M P_{ci} + \alpha_c + \alpha_i + \varepsilon_{ci} \tag{13}$$

 ΔR_{ci} is the country-industry level change in the stock of robots. I measure the change both between 2005 and 2014, and 1993 and 2004. I use the first version as baseline and the second one as placebo. $\Delta M P_{ci}$ is the country-industry level change in multinational production between 2004 and 2015. Finally, α_c and α_i are country-level, respectively industry-level, fixed effects to account for the fact that different countries and industries have experiences heterogeneous trends robot adoption trends.

The main parameter of interest is β . I standardize all variables to have zero mean in the sample before performing the estimation. Therefore, it measures the effect of increasing the change in multinational production by one unit on the change in the robot stock at the country-industry level. Since I compute ΔMP_{ci} between 2005 and 2014, I expect β to be positive and significant when using ΔR_{ci} between 2005 and 2014 as an outcome, and insignificant when using the change between 1994 and 2004.

Table B.3 shows the results. As expected, the total change in multinational production positively correlates with the total change in the number of deployed robot during the sample period, but not before.

Dependent Variables:	$\begin{array}{c} \Delta \text{Robots}_{ci} \ (2005\text{-}2014) \\ (1) \end{array}$	$\begin{array}{c} \Delta \text{Robots}_{ci} \ (1994\text{-}2004) \\ (2) \end{array}$
Δ Multinational Production _{ci}	0.29^{*}	0.03
	(0.17)	(0.04)
Country FE	Yes	Yes
Industry FE	Yes	Yes
Observations	740	740
Estimator	OLS	OLS

Table B.3: CROSS-COUNTRY INDUSTRY-LEVEL CORRELATIONS (LONG-DIFFERENCE)

Note: $\Delta \text{Robots}_{ci}$ (2005-2014) is the robot stock change in country *c*'s industry *i* between 2005 and 2014. $\Delta \text{Robots}_{ci}$ (1994-2005) is the robot stock change in country *c*'s industry *i* between 1994 and 2004. $\Delta \text{Multinational Production}_{ci}$ is the change in the gross output produced by affiliates of foreign firms in country *c*'s industry *i* between 2005 and 2014. Heteroscedasticity-robust standard errors in parenthesis. Significance levels: *** 0.01, ** 0.05, * 0.1.

B.4 Figures



Figure B.1: Correlation between Multinational Production and Robots

Note: The Figure shows the correlation between the log of multinational production (MP) and the log of the number of deployed robots by country. I compute correlation across industries and years within countries. The correlation is positive in most countries. The sample average is about 30%, and it is strongest in China, where it peaks at about 60%.

Figure B.2: Robots per Thousand Employees vs. Multinational Production



Note: The Figure shows the linear fit I obtain by regressing the log of the number of robots per thousand employees on the log of multinational production at the country-industry-year level. The estimated coefficient is about 0.5% and is significant at the 1% level.

Appendix C Additional Firm-Level Results

C.1 Tables

Dependent Variable:					Foreign-	Owned _{ft}				
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
Interecept	-2.6***	-2.6***	-7.4***	-5.7***	-6.0***	-6.1***	-6.1***	-5.7***	-5.3***	-5.4***
	(0.04)	(0.04)	(0.36)	(0.47)	(0.49)	(0.54)	(0.53)	(0.55)	(0.66)	(0.66)
Sales Growth Rate_{ft}		0.25***	0.44***	0.30***	0.26***	0.27***	0.26***	0.25***	0.25***	0.25***
		(0.07)	(0.08)	(0.08)	(0.09)	(0.09)	(0.09)	(0.09)	(0.09)	(0.09)
$Log(Sales)_{ft-1}$			(0.02)	$(0.13)^{(0.04)}$	(0.07)	(0.08)	(0.07)	(0.07)	(0.03)	(0.03)
Log(Employees) et 1			(0.02)	(0.04) 0.25^{***}	(0.00) 0.29^{***}	(0.00) 0.29^{***}	(0.07) 0.28^{***}	(0.07) 0.24^{***}	(0.08) 0.25^{***}	(0.08) 0.25^{***}
108(1111)105000)J1=1				(0.06)	(0.07)	(0.07)	(0.07)	(0.08)	(0.08)	(0.08)
$Log(Wages)_{ft-1}$				· · ·	0.32^{**}	0.32^{**}	0.31^{*}	0.23	0.25	0.27
					(0.16)	(0.16)	(0.17)	(0.17)	(0.17)	(0.18)
$Log(Investment)_{ft-1}$						-0.005	-0.005	-0.008	-0.009	-0.009
						(0.01)	(0.01)	(0.01)	(0.01)	(0.01)
$Log(Physical Inputs)_{ft-1}$							(0.01)	(0.01)	(0.008)	(0.01)
Log(B&D Expenses) et a							(0.05)	0.03***	0.02***	0.02***
$\log(100D \text{ Expenses}) ft-1$								(0.007)	(0.002)	(0.008)
$Log(Exports)_{ft-1}$								()	0.02^{*}	0.02**
									(0.01)	(0.01)
No. Export $Countries_{ft-1}$										-0.05
										(0.04)
Observations	18,896	$18,\!896$	$18,\!896$	$18,\!896$	$18,\!896$	$18,\!896$	$18,\!896$	$18,\!896$	$18,\!896$	$18,\!896$
Estimator	Probit	Probit	Probit	Probit	Probit	Probit	Probit	Probit	Probit	Probit

Table C.1: PROPENSITY SCORE (PROBIT) REGRESSIONS

Note: The unit of observation is a firm. Foreign-Owned_{ft} is a binary variable equal to 1 if firm f becomes foreign-owned at some point during its life and 0 otherwise. Sales Growth Rate_{ft} is the sales' growth rate between the acquisition year and the one before. $\operatorname{Log}(\operatorname{Sales})_{ft-1}$ is the lag of (the log of) firm sales. $\operatorname{Log}(\operatorname{Employees})_{ft-1}$ is the lag of (the log of) the number of employees. $\operatorname{Log}(\operatorname{Wages})_{ft-1}$ is the lag of (the log of) firm wages. $\operatorname{Log}(\operatorname{Investment})_{ft-1}$ is the lag of (the log of) firm investment. $\operatorname{Log}(\operatorname{Physical Inputs})_{ft-1}$ is the lag of (the log of) firm physical inputs. $\operatorname{Log}(\operatorname{R\&D} \operatorname{Expenses})_{ft-1}$ is the lag of (the log of) R\&D expenses. $\operatorname{Log}(\operatorname{Export})_{ft-1}$ is the lag of (the log of) the number of export countries served by the firm. Heteroscedasticity-robust standard errors in parenthesis. Significance levels: *** 0.01, ** 0.05, * 0.1.

Sample	Employees	Sales	Export Value	No. Export Markets	Wages	Physical Inputs	Investment	R&D Expenses	Sales' Growth
Unweighed	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00
Reweighed	0.82	1.00	0.74	0.73	0.94	0.97	0.74	0.96	0.48

Table C.2: COVARIATES' BALANCE BEFORE AND AFTER REWEIGHING

Note: Each row shows the p-value associated with the null hypothesis that acquired and non-acquired firms have the same average characteristics along the dimensions reported in columns. The first row shows the p-value computed in the unmatched (baseline) sample. The second row shows the p-value computed in the matched sample, i.e., after reweighing acquired firms by $1/\hat{p}$ and non-acquired ones by $1/(1-\hat{p})$, being \hat{p} the estimated probability of becoming foreign-owned. Whereas I always reject the null hypothesis that the two groups have different average characteristics, I never reject it post-matching.

Dependent Variable:			$\operatorname{Robot}_{ft}$	
	(1)	(2)	(3)	(4)
Foreign-Owned _{ft} × Diffusion Tercile = 1	0.27***			
·	(0.09)			
Foreign-Owned _{<i>tt</i>} × Diffusion Tercile = 2	0.13^{**}			
.	(0.06)			
Foreign-Owned _{<i>tt</i>} × Diffusion Tercile = 3	-0.02			
- ,,-	(0.06)			
Foreign-Owned $_{ft}$		0.13^{*}	0.15^{**}	-0.06
- ,,-		(0.08)	(0.06)	(0.06)
Firm FE	Yes	Yes	Yes	Yes
Industry-Year FE	Yes	Yes	Yes	Yes
Sample	Full	Diffusion Tercile $= 1$	Diffusion Tercile $= 2$	Diffusion Tercile $= 3$
Observations	18,520	1,592	11,775	5,153
Estimator	IPRW	IPRW	IPRW	IPRW

Table C.3: INDUSTRY-LEVEL HETEROGENEITY

Note: The unit of observation is a firm-year pair. Robot_{ft} is a binary variable equal to 1 if firm f employs a robot in year t and 0 otherwise. Foreign-Owned_{ft} is a binary variable equal to 1 if firm f is foreign-owned in year t and 0 otherwise. I construct Robots' Diffusion Tercile as follows. First, I compute the number of industrial robots deployed in each Spanish manufacturing industry in 1993. Second, I group industries into terciles. The first tercile includes industries with the lowest robot adoption rate in 1993. In all columns, I reweigh acquired firms by $1/\hat{p}$ and non-acquired ones by $1/(1-\hat{p})$, being \hat{p} the estimated probability of becoming foreign-owned. Cluster standard errors by firm in parenthesis. Significance levels: *** 0.01, ** 0.05, * 0.1.

Dependent Variable:		$\operatorname{Robot}_{ft}$	
	(1)	(2)	(3)
Exp. via Foreign $\operatorname{Parent}_{ft}$	0.30***	0.45^{**}	0.46**
<i>y</i> -	(0.08)	(0.19)	(0.19)
Foreign-Owned $_{tt-1}$	0.29**		0.28**
	(0.14)		(0.14)
Exp. via Foreign Parent _{<i>tt</i>} × Foreign-Owned _{<i>tt</i>-1} = 1	-0.37*		-0.40**
	(0.20)		(0.20)
$\operatorname{Exporter}_{tt-1}$		0.14^{**}	0.13**
		(0.06)	(0.05)
Exp. via Foreign $\operatorname{Parent}_{ft} \times \operatorname{Exporter}_{ft-1} = 1$		-0.29	-0.12
		(0.20)	(0.19)
Firm FE	Yes	Yes	Yes
Industry-Year FE	Yes	Yes	Yes
Observations	2,895	2,895	2,895
Estimator	IPRW	IPRW	IPRW

Table C.4: FIRM-LEVEL MECHANISM REGRESSIONS (STEP TWO, HETEROGENEITY)

Note: The unit of observation is a firm-year pair. Exp. via Foreign $\operatorname{Parent}_{ft}$ is binary variable equal to 1 if firm f exports via its foreign parental network at time t and 0 otherwise. Foreign-Owned_{ft} is binary variable equal to 1 if firm f is foreign-owned at time t and 0 otherwise. Exporter_{ft} is binary variable equal to 1 if firm f exports at time t and 0 otherwise. In all columns, I reweigh acquired firms by $1/\hat{p}$ and non-acquired ones by $1/(1-\hat{p})$, being \hat{p} the estimated probability of becoming foreign-owned. Cluster standard errors by firm in parenthesis. Significance levels: *** 0.01, ** 0.05, * 0.1.

Table C.5: COVARIATES' BALANCE BEFORE AND AFTER REWEIGHING (DOMESTIC ACQUISITIONS)

Sample	Employees	Sales	Wages	Physical Inputs	Exporter Status	Sales Growth	Employment Growth
Unweighed	0.00	0.00	0.00	0.00	0.00	0.00	0.00
Reweighed	0.70	0.84	0.96	0.82	0.72	0.79	0.90

Note: Each row shows the p-value associated with the null hypothesis that acquired and non-acquired firms have the same average characteristics along the dimensions reported in columns. The first row shows the p-value computed in the unmatched (baseline) sample. The second row shows the p-value computed in the matched sample, i.e., after reweighing acquired firms by $1/\hat{p}$ and non-acquired ones by $1/(1-\hat{p})$, being \hat{p} the estimated probability of being acquired by a domestic firm. Whereas I always reject the null hypothesis that the two groups have different average characteristics, I never reject it post-matching.

Dependent Variable:	$\operatorname{Robot}_{ft}$				
	(1)	(2)	(3)	(4)	
$Merger_{ft}$	0.13^{*}	0.20***	0.19***	0.15	
5	(0.07)	(0.07)	(0.07)	(0.10)	
Firm FE	No	Yes	Yes	Yes	
Industry-Year FE	No	Yes	Yes	Yes	
Selection Controls	No	No	Yes	No	
Observations	16,264	16,264	15,377	16,264	
Estimator	OLS	OLS	OLS	IPRW	

Table C.6: FIRM-LEVEL REGRESSIONS (DOMESTIC ACQUISITIONS)

Note: The unit of observation is a firm-year pair. $\operatorname{Robot}_{ft}$ is a binary variable equal to 1 if firm f employs a robot in year t and 0 otherwise. Merger_{ft} is a binary variable equal to 1 if firm f was acquired by another domestic firm in year $k \leq t$ and 0 otherwise. In the last column, I reweigh acquired firms by $1/\hat{p}$ and non-acquired ones by $1/(1-\hat{p})$, being \hat{p} the estimated probability of being acquired by a domestic firm. Cluster standard errors by firm in parenthesis. Significance levels: *** 0.01, ** 0.05, * 0.1.

Dependent Variables:	Batch Manuf. _{ft} (1)	$\begin{array}{c} \text{Mass Manuf.}_{ft} \\ (2) \end{array}$	Mixed Manuf. _{ft} (3)	Continuous Manuf. _{ft} (4)
$\operatorname{Foreign-Owned}_{ft}$	-0.08^{**} (0.04)	$0.02 \\ (0.05)$	-0.006 (0.009)	0.06^{*} (0.04)
Firm FE	Yes	Yes	Yes	Yes
Industry-Year FE	Yes	Yes	Yes	Yes
Observations Estimator	18,214 IPRW	18,214 IPRW	18,214 IPRW	18,214 IPRW

Table C.7: Type of Production

Note: The unit of observation is a firm-year pair. Batch $\operatorname{Manuf}_{ft}$ is a binary variable equal to 1 if firm f does batch manufacturing in year t and 0 otherwise. Mass $\operatorname{Manuf}_{ft}$ is a binary variable equal to 1 if firm f does mass manufacturing in year t and 0 otherwise. Mixed $\operatorname{Manuf}_{ft}$ is a binary variable equal to 1 if firm f does mixed manufacturing in year t and 0 otherwise. Continuous $\operatorname{Manuf}_{ft}$ is a binary variable equal to 1 if firm f does continuous manufacturing in year t and 0 otherwise. Foreign-Owned $_{ft}$ is a binary variable equal to 1 if firm f does continuous manufacturing in year t and 0 otherwise. Foreign-Owned $_{ft}$ is a binary variable equal to 1 if firm f is foreign-owned in year t and 0 otherwise. In all columns, I reweigh acquired firms by $1/\hat{p}$ and non-acquired ones by $1/(1-\hat{p})$, being \hat{p} the estimated probability of becoming foreign-owned. Cluster standard errors by firm in parenthesis. Significance levels: *** 0.01, ** 0.05, * 0.1.

Dependent Variables:	CAD Manufacturing $_{ft}$	Numerically Ctrl Machines $_{ft}$	Flex. Systems_{ft}	Any Non-Robotic Technology $_{ft}$
	(1)	(2)	(3)	(4)
Foreign-Owned $_{ft}$	0.05^{*}	-0.06	0.02	0.003
	(0.03)	(0.04)	(0.03)	(0.04)
Firm FE	Yes	Yes	Yes	Yes
Industry-Year FE	Yes	Yes	Yes	Yes
Observations	$18,\!467$	18,467	18,463	18,467
Estimator	IPRW	IPRW	IPRW	IPRW

Table C.8: OTHER AUTOMATION-TYPE NON-ROBOTIC TECHNOLOGY

Note: The unit of observation is a firm-year pair. CAD Manufacturing_{ft} is a binary variable equal to 1 if firm f used CAD manufacturing in year t and 0 otherwise. Numerically Ctrl Machines_{ft} is a binary variable equal to 1 if firm f uses numerically controlled machines in year t and 0 otherwise. Flex. Systems_{ft} is a binary variable equal to 1 if firm f employs flexible systems in year t and 0 otherwise. Any Non-Robotic Technology_{ft} is a binary variable equal to 1 if firm f uses any of these three technologies and 0 otherwise. Foreign-Owned_{ft} is a binary variable equal to 1 if firm f is foreign-owned in year t and 0 otherwise. In all columns, I reweigh acquired firms by $1/\hat{p}$ and non-acquired ones by $1/(1-\hat{p})$, being \hat{p} the estimated probability of becoming foreign-owned. Cluster standard errors by firm in parenthesis. Significance levels: *** 0.01, ** 0.05, * 0.1.

Dependent Variables:	$Log(Sales)_{ft}$	$Log(Employees)_{ft}$	$Log(Wages)_{ft}$	$Log(Labor Productivity)_{ft}$	$Log(Sales/Wages)_{ft}$
	(1)	(2)	(3)	(4)	(5)
Foreign-Owned $_{ft}$	0.32***	0.20***	0.09***	0.12^{***}	0.23***
U	(0.06)	(0.07)	(0.03)	(0.04)	(0.05)
Firm FE	Yes	Yes	Yes	Yes	Yes
Industry-Year FE	Yes	Yes	Yes	Yes	Yes
Observations	20,529	20,530	20,530	20,529	20,529
Estimator	IPRW	IPRW	IPRW	IPRW	IPRW

Table C.9: WITHIN-FIRM GAINS AND DISTRIBUTIONAL OUTCOMES

Note: The unit of observation is a firm-year pair. $\text{Log}(\text{Sales})_{ft}$ is the log of firm sales. $\text{Log}(\text{Employees})_{ft}$ is the log of the number of employees. $\text{Log}(\text{Wages})_{ft}$ is the log of firm wages. $\text{Log}(\text{Labor Productivity})_{ft}$ is the log of the ratio between firm sales and number of employees. $\text{Log}(\text{Sales}/\text{Wages})_{ft}$ is the ratio between firm sales and wages. Foreign-Owned $_{ft}$ is a binary variable equal to 1 if firm f is foreign-owned in year t and 0 otherwise. In all columns, I reweigh acquired firms by $1/\hat{p}$ and non-acquired ones by $1/(1-\hat{p})$, being \hat{p} the estimated probability of becoming foreign-owned. Cluster standard errors by firm in parenthesis. Significance levels: *** 0.01, ** 0.05, * 0.1.

C.2 Figures



Figure C.1: EVENT-STUDY (ALTERNATIVE SAMPLE)

Note: The Figure plots the estimates (and the 95% confidence intervals around them) that I obtain from equation (2). The unit of observation is a firm-year pair, and I cluster standard errors by firm. The sample includes both acquired and always domestic firms. Starting from the first year after the acquisition date, affiliates become about 20% more likely to employ robots than in the year before the acquisition.



Figure C.2: EVENT-STUDY (SUN AND ABRAHAM, 2021)

Note: The Figure plots the estimates (and the 95% confidence intervals around them) that I obtain from equation (2) using the method proposed by Sun and Abraham (2021). The unit of observation is a firm-year pair, and I cluster standard errors by firm. The sample includes both acquired and always domestic firms. Starting from second first year after the acquisition date, affiliates become about 25% more likely to employ robots than in the year before the acquisition. The estimates are not statistically different from those in Figure C.1



Figure C.3: EVENT-STUDY (PROPENSITY SCORE REWEIGHING)

Note: The Figure plots the estimates (and the 95% confidence intervals around them) that I obtain from equation (2) after reweighing acquired firms by $1/\hat{p}$ and non-acquired ones by $1/(1-\hat{p})$, being \hat{p} the estimated probability of becoming foreign-owned. The unit of observation is a firm-year pair, and I cluster standard errors by firm. Starting from the first year after the acquisition date, affiliates become about 20% more likely to employ robots than comparable domestic-owned firms. Whereas it deliver similar post-acquisition estimates as in Figure C.1, the propensity reweighing procedure further assuages pre-acquisition trends.



Figure C.4: EVENT-STUDY (EXPORT VIA FOREIGN PARENT)

Note: The Figure plots the estimates (and the 95% confidence intervals around them) that I obtain by estimating equation (2) using the binary variable equal to 1 if a firm exports via her foreign parental network as outcome variable. I reweigh acquired firms by $1/\hat{p}$ and non-acquired ones by $1/(1-\hat{p})$, being \hat{p} the estimated probability of becoming foreign-owned. The unit of observation is a firm-year pair, and I cluster standard errors by firm. Starting from the first year after the acquisition date, affiliates are about 50% to export via their foreign parents than in the year before the acquisition.

Figure C.5: EVENT-STUDY (EXPORTS VALUE)



Note: The Figure plots the estimates (and the 95% confidence intervals around them) that I obtain by estimating equation (2) using the log of the exports as outcome variable. I reweigh acquired firms by $1/\hat{p}$ and non-acquired ones by $1/(1-\hat{p})$, being \hat{p} the estimated probability of becoming foreign-owned. The unit of observation is a firm-year pair, and I cluster standard errors by firm. Starting from the first year after the acquisition date, affiliates increase their exports by about 30% compared to the year before the acquisition.



Figure C.6: COVARIATES' DENSITY PLOTS BY OWNERSHIP

Note: This Figure shows the empirical probability density function (pdf) of the log of employees, sales, physical inputs, R&D expenses, wages, and investment by ownership type. I estimate the empirical pdf functions for domestic-owned firms based on their lifetime characteristics. I estimate it only for the years before the acquisition date for those acquired by another domestic firm. Overall, to-be-acquired firms outperform always domestic-owned ones along all dimensions already before the acquisition.



Figure C.7: EVENT STUDY (DOMESTIC ACQUISITIONS)

Note: The Figure plots the estimates (and the 95% confidence intervals around them) that I obtain from equation (2). The only difference with the baseline specification is that the O_{ft}^s dummy(-ies) identifies the periods before or after the domestic acquisition date. The unit of observation is a firm-year pair, and I cluster standard errors by firm. The Figure does not deliver any significant effect of being acquired by a Spanish firm on the probability of robot adoption post-acquisition.

Appendix D Additional Model-Related Results

D.1 Figures

Figure D.1: GMM SENSITIVITY



Note: The Figure shows the sensitivity of σ , β_1 and β_2 in equation (10) to a one-unit of (excluded) instrument standard deviation violation of the moment condition $\mathbb{E}[L_{ft-1}u_{ft}] = 0$. I compute sensitivity using the method of Andrews et al. (2017). Overall, the three estimated coefficients are insensitive to local deviations in this moment condition.

D.2 Counterfactual Scenarios Implementation

I study how the total change in industry-level TFP and labor share (in cost) in the Spanish manufacturing industry over the sample period would have looked like absent robot adoption and/or foreign ownership. To do so, I consider the following system of difference equations based on equations (10) and (12):

$$\log(\widehat{B}_{ft}) = \log(\widehat{B}_{ft-1}) + \widehat{\beta}_1 R_{ft-1} + \widehat{u}_{ft};$$
(14)

$$\log(\widehat{z}_{ft}) = \widehat{\zeta}_0 + \widehat{\zeta}_1 \log(\widehat{z}_{ft-1}) + \widehat{\zeta}_2 R_{ft-1} + \widehat{\zeta}_3 O_{ft-1} + \widehat{\xi}_{ft}.$$
 (15)

being $\log(\widehat{B}_{ft-1}) = \sum_{s=0}^{4} \left(\log\left(\frac{M_{ft-1}}{L_{ft-1}}\right) - \widehat{\sigma} \log\left(\frac{w_{ft-1}}{r_{jt-1}}\right) \right)^s$. Notice that I set $\widehat{\beta}_2 = 0$ in equation (10) because it is not significant in Table 5. I simulate two different scenarios. In the first one, I shut down the effect of foreign ownership on productivity. Second, I shut down both the effect of foreign ownership and that of robot adoption on productivity. In each scenario, I simulate counterfactual outcomes in two steps. First, using the initial (observed) values of total factor and labor-augmenting productivity, I compute the counterfactual TFP and labor share at the firm-year level. Second, I aggregate the results at the industry-year level using the observed firms' employment shares each year, which I keep constant across scenarios. I define industry-level TFP and the industry-level labor share as follows:

$$Z_{t} = \sum_{f} \omega_{ft} \log(\widehat{z}_{ft}); \qquad (16)$$

$$LS_{t} = \sum_{f} \omega_{ft} \frac{w_{ft}L_{ft}}{w_{ft}L_{ft} + r_{ft}M_{ft}}$$

$$= \sum_{f} \omega_{ft} \frac{1}{1 + \frac{w_{ft}L_{ft}}{r_{ft}M_{ft}}}$$

$$= \sum_{f} \omega_{ft} \frac{1}{1 + \left(\frac{w_{ft}}{r_{jt}}\right)^{\widehat{\sigma}-1} \frac{1}{\widehat{B}_{ft}}}. \qquad (17)$$

being ω_{ft} the employment share of firm f in year t in the sub-sample of always active firms. When computing the labor share, I use the relationship implied by the first-order conditions governing the firm-level optimal input mix in equation (8) and hold w_{ft}/r_{jt} at its observed value in the data and simulate \hat{B}_{ft} using equation (14).

First Scenario - No MNEs

• Step 1. I discount the robot adoption coefficients $\hat{\beta}_1$ in equation (14) and $\hat{\zeta}_2$ in equation (15). Moreover, I set the effect of foreign ownership on productivity

 $\widehat{\zeta}_3 = 0$ in equation (15):

$$\log(\widehat{B}_{ft}) = \log(\widehat{B}_{ft-1}) + \widehat{\beta}_1 (1 - \widehat{\theta} \times O_{ft-1}) R_{ft-1} + \widehat{u}_{ft};$$
(18)

$$\log(\widehat{z}_{ft}) = \widehat{\zeta}_0 + \widehat{\zeta}_1 \log(\widehat{z}_{ft-1}) + \widehat{\zeta}_2 (1 - \widehat{\theta} \times O_{ft-1}) R_{ft-1} + \widehat{\xi}_{ft}.$$
 (19)

The terms $\hat{\beta}_1(1-\hat{\theta}\times O_{ft-1})R_{ft-1}$ and $\hat{\zeta}_2(1-\hat{\theta}\times O_{ft-1})R_{ft-1}$ capture the indirect effect that foreign-ownership exerts on robot adoption. I use $\hat{\theta} = 10\%$ from Table 2. Therefore, if $O_{ft-1} = R_{ft-1} = 1$, I discount the foreign-ownership robot adoption premium $\hat{\theta}$ from the effects of robot adoption on both productivities (i.e., $\hat{\beta}_1$ and $\hat{\zeta}_2$). Remember that I bin the data into seven mutually exclusive time intervals: [1993-1995], [1996-1998], [1999-2001], [2002-2004], [2005-2007], [2008-2010], and [2011-2014]. Because I include one-year lags in equation (18), I can only recover \hat{u}_{ft} from [1996-1998] onward. Moreover, because I infer \hat{z}_{ft} from equation (11) and include one-year lags also in equation (19), I can only recover $\hat{\xi}_{ft}$ from 1998 onward. I simulate equations (18) and (19) forward for [1999-2001], [2002-2004], [2005-2007], [2008-2010], and [2011-2014]. Notice that the effect of shutting down robot adoption compounds over time due to the first-order autoregressive structure of equations (18) and (19).

• Step 2. I compute industry-level TFP using equation (16) and the industry-level labor share using equation (17).

Second Scenario - Neither MNEs nor Robots I modify the first-step equations (18) and (19) as:

$$\log(\widehat{B}_{ft}) = \log(\widehat{B}_{ft-1}) + \widehat{u}_{ft}; \tag{20}$$

$$\log(\widehat{z}_{ft}) = \widehat{\zeta}_0 + \widehat{\zeta}_1 \log(\widehat{z}_{ft-1}) + \widehat{\xi}_{ft}.$$
(21)

Then, I follow the same steps as in the first scenario.
CENTRE FOR ECONOMIC PERFORMANCE Recent Discussion Papers

1853	Andrew E. Clark Conchita D'Ambrosio Niccoló Gentile Alexandre Tkatchenko	What makes a satisfying life? Prediction and interpretation with machine-learning algorithms
1852	Xiang Ding Teresa C. Fort Stephen J. Redding Peter K. Schott	Structural change within versus across firms: evidence from the United States
1851	Christos Genakos Eleni Kyrkopoulou	Social policy gone bad educationally: unintended peer effects from transferred students
1850	Antonin Bergeaud Cyril Verluise	A new dataset to study a century of innovation in Europe and in the US
1849	Jo Blanden Mattias Doepke Jan Stuhler	Education inequality
1848	Martina Manara Tanner Regan	Ask a local: improving the public pricing of land titles in urban Tanzania
1847	Rebecca Freeman Kalina Manova Thomas Prayer Thomas Sampson	Unravelling deep integration: UK trade in the wake of Brexit
1846	Nicholas Bloom Scott W. Ohlmacher Cristina J. Tello-Trillo Melanie Wallskog	Pay, productivity and management
1845	Martin Gaynor Adam Sacarny Raffaella Sadun Chad Syverson Shruthi Venkatesh	The anatomy of a hospital system merger: the patient did not respond well to treatment

1844	Tomaz Teodorovicz Raffaella Sadun Andrew L. Kun Orit Shaer	How does working from home during Covid- 19 affect what managers do? Evidence from time-use studies
1843	Giuseppe Berlingieri Frank Pisch	Managing export complexity: the role of service outsourcing
1842	Hites Ahir Nicholas Bloom Davide Furceri	The world uncertainty index
1841	Tomaz Teodorovicz Andrew L. Kun Raffaella Sadun Orit Shaer	Multitasking while driving: a time use study of commuting knowledge workers to access current and future uses
1840	Jonathan Gruber Grace Lordan Stephen Pilling Carol Propper Rob Saunders	The impact of mental health support for the chronically ill on hospital utilisation: evidence from the UK
1839	Jan Bietenbeck Andreas Leibing Jan Marcus Felix Weinhardt	Tuition fees and educational attainment
1838	Jan De Loecker Tim Obermeier John Van Reenen	Firms and inequality
1837	Ralph De Haas Ralf Martin Mirabelle Muûls Helena Schweiger	Managerial and financial barriers during the green transition
1836	Lindsay E. Relihan	Is online retail killing coffee shops? Estimating the winners and losers of online retail using customer transaction microdata

The Centre for Economic Performance Publications Unit Tel: +44 (0)20 7955 7673 Email <u>info@cep.lse.ac.uk</u> Website: <u>http://cep.lse.ac.uk</u> Twitter: @CEP_LSE